## UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

#### FORM 8-K

## CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of report (Date of earliest event reported): February 9, 2011

Tegal Corporation (Exact name of Registrant as Specified in its Charter)

Delaware (State or other jurisdiction of incorporation) 000-26824 (Commission File Number) 68-0370244 (I.R.S. Employer Identification No.)

2201 South McDowell Boulevard Petaluma, CA 94954 (Address of Principal Executive Offices)

(707) 763-5600 (Registrant's telephone number, including area code)

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

□ Written communication pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

## ITEM 1.01. Entry Into a Material Definitive Agreement.

On February 9, 2011, Tegal Corporation, a Delaware corporation (the "<u>Company</u>"), and SPP Process Technology Systems Limited, a company incorporated and registered in England and Wales (together with its subsidiary designees, "<u>Purchaser</u>"), entered into an Asset Purchase Agreement (the "<u>Purchase Agreement</u>") pursuant to which the Company sold to Purchaser (the "<u>Disposition</u>") all of the shares of Tegal France, SAS, the Company's wholly-owned subsidiary (the "<u>Tegal France Shares</u>"), and product lines and certain equipment, intellectual property and other assets relating to the Company's Deep Reactive Ion Etch plasma etch systems and certain related technology (together with the Tegal France Shares, the "<u>Purchased Assets</u>"). The Purchaser also assumed existing customer contracts, including all installation and warranty obligations of existing customers, and other liabilities arising after the closing of the Disposition (the "<u>Assumed Liabilities</u>").

In connection with the Disposition, the Company and the Purchaser entered into related agreements for the transfer and licensing of patents, trademarks and other intellectual property associated with the Included Businesses, including a royalty-free Trademark License Agreement allowing for the limited use of the Tegal trademark by the Purchaser solely in connection with future sales related to the Included Businesses and solely in combination with the trademarks transferred to the Purchaser, as well as written assignments to the Purchaser of all rights in the patents and trademarks that are part of the Disposition.

The Disposition closed immediately after execution of the Purchase Agreement. The consideration paid by the Purchaser for the Disposition totaled approximately \$2.1 million, comprised of approximately \$0.5 million of Assumed Liabilities and \$1.6 million in cash, of which \$200,000 in cash will be held in escrow for one year after the closing of the Disposition to satisfy any indemnification obligations of the Company under the Purchase Agreement.

The descriptions of the Purchase Agreement and the Trademark License Agreement provided above are qualified in their entirety by reference to the full text of such agreements, copies of which have been filed as Exhibits 10.1 and 10.2, respectively, to this report and are incorporated herein by reference.

#### ITEM 2.01. Completion of Acquisition or Disposition of Assets.

On February 9, 2011, the Company completed the Disposition. The information in Item 1.01 above is incorporated herein by reference. On February 9, 2011, the Company issued a press release regarding the Disposition, a copy of which is attached as Exhibit 99.1 hereto.

### ITEM 9.01. Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description
10.1	Asset Purchase Agreement, dated as of February 9, 2011, between SPP Process Technology Systems Ltd. and Tegal Corporation.

10.2 Trademark Assignment Agreement, dated as of February 9, 2011, between Tiger Corporation and SPP Technology Systems UK Limited.

<sup>99.1</sup> Press release issued by Tegal Corporation, dated February 9, 2011.

## SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: February 15, 2011

## TEGAL CORPORATION

By:/s/ Christine T. HergenrotherName:Christine T. HergenrotherTitle:Chief Financial Officer

# EXHIBIT INDEX

Exhibit No.	Description
10.1	Asset Purchase Agreement, dated February [9], 2011, between SPP Process Technology Systems Ltd. and Tegal Corporation.
10.2	Trademark Assignment Agreement, dated February [9], 2011, between Tiger Corporation and SPP Technology Systems UK Limited.
99.1	Press release issued by Tegal Corporation, dated February [9], 2011.

# **EXECUTION VERSION**

## ASSET PURCHASE AGREEMENT

between

# SPP PROCESS TECHNOLOGY SYSTEMS LTD.

and

TEGAL CORPORATION

February 9, 2011

## TABLE OF CONTENTS

	Page
ARTICLE I DEFINITIONS AND INTERPRETATION	1
1.1 Definitions	1
1.2 Other Definitional and Interpretive Matters	9
ARTICLE II PURCHASE AND SALE OF ASSETS	10
2.1 Purchased Assets	10
2.2 Excluded Assets	11
2.3 Assumed Liabilities	12
2.4 Excluded Liabilities	13
2.5 <u>Purchase Price</u>	14
2.6 <u>Tegal France</u>	15
2.7 <u>Nonassignable Assets</u>	16
2.8 <u>Allocation of the Purchase Price</u>	17
2.9 Bulk Transfer Laws	17
ARTICLE III THE CLOSING	18
3.1 <u>Closing Date</u>	18
3.2 <u>Deliveries</u>	18
3.3 <u>Physical Delivery of Tangible Assets</u>	19
3.4 <u>Withholding Taxes</u>	19
ARTICLE IV REPRESENTATIONS AND WARRANTIES OF SELLER	20
4.1 Organization and Power	20
4.2 <u>Authorization; Enforceability</u>	20
4.3 <u>Subsidiaries</u>	21
4.4 Consents and Approvals; No Violations	22
4.5 <u>Title to Properties; Good Title Conveyed</u>	22
4.6 <u>Financial Statements</u>	23
4.7 Intellectual Property	23
4.8 <u>Material Contracts</u>	25
4.9 <u>Litigation</u>	26
4.10 Compliance with Laws; Permits	26
4.11 <u>Real Property</u>	26
4.12 <u>Taxes; Tax Returns</u>	27

## TABLE OF CONTENTS (continued)

	Page
4.13Books and Records	27
4.14 <u>Accounts Receivable</u>	28
4.15 <u>Inventory</u>	28
4.16Warranty Claims	28
4.17 <u>Product Liability</u>	28
4.18 <u>Insurance</u>	28
4.19 <u>Employees</u>	29
4.20 <u>Employee Benefits</u>	30
4.21 <u>Environmental Matters</u>	31
4.22 <u>Changes Since Balance Sheet Date</u>	31
4.23 <u>Financial Advisors</u>	32
ARTICLE V REPRESENTATIONS AND WARRANTIES OF PURCHASER	32
5.1 Organization	32
5.2 Authorization; Enforceability	32
5.3 Consents and Approvals; No Violations	32
5.4 Litigation	33
5.5 Financial Advisors	33
ARTICLE VI COVENANTS AND AGREEMENTS	33
6.1 <u>Publicity</u>	33
6.2 <u>Tax Matters</u>	33
6.3 <u>Further Assurances</u>	36
6.4 <u>Business Records</u>	36
6.5 Employment Matters	37
6.6 Miscellaneous Transition Matters	37
6.7 <u>Non-Competition</u>	38
6.8 Non-Solicitation	38
6.9 <u>Alcatel Consent</u>	38
6.10 <u>Rudolph Consent</u>	38
ARTICLE VII INDEMNITY	39
7.1 <u>Survival</u>	39
7.2 Indemnification by Seller	39

## TABLE OF CONTENTS (continued)

	-	Page
7.3	Indemnification by Purchaser	39
7.4	Limitations on Indemnification	40
7.5	Third-Party Claims	41
7.6	Escrow	41
7.7	Tax Effect on Indemnification Payments	42
7.8	Exclusive Remedy	42
ARTICLE VII	I MISCELLANEOUS	42
8.1	Expenses	42
8.2	Notices	42
8.3	Counterparts	43
8.4	Section Headings	43
8.5	Amendments; No Waivers	43
8.6	Entire Agreement; No Assignment	44
8.7	No Third Party Beneficiaries	44
8.8	Governing Law	44
8.9	Severability	44
8.10	Specific Performance: Preservation of Remedies	44

# <u>Exhibits</u>

- A Form of Escrow Agreement
- B Bill of Sale
- C Assignment and Assumption Agreement
- D Trademark License Agreement



This ASSET PURCHASE AGREEMENT is entered into as of February 9, 2011 (the "<u>Effective Date</u>") by SPP Process Technology Systems Ltd., a company incorporated and registered in England and Wales ("<u>Purchaser</u>"), and Tegal Corporation, a Delaware corporation ("<u>Seller</u>"). Purchaser and Seller are sometimes referred to herein collectively as the "<u>Parties</u>" and individually as a "<u>Party</u>."

A. Seller is engaged in the following businesses, among others: (a) the development, design, manufacture, marketing and servicing of its "Deep Reactive Ion Etch" plasma etch systems and related Intellectual Property (the "<u>DRIE Business</u>"); (b) the development and commercial exploitation of its Compact<sup>TM</sup> cluster platform and related Intellectual Property (the "<u>Compact Business</u>"); and (c) the development and commercial exploitation of its low-temperature, single-wafer parylene deposition Intellectual Property (the "<u>Pluto Business</u>") and, together with the DRIE Business and the Compact Business, the "<u>Included Businesse</u>"); and

B. On the terms and subject to the conditions of this Agreement, Purchaser desires to purchase and acquire from Seller, and Seller desires to sell and transfer to Purchaser the Purchased Assets (as defined below);

NOW, THEREFORE, in consideration of respective covenants, representations, warranties and agreements herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties hereby agree as follows:

## ARTICLE I

#### DEFINITIONS AND INTERPRETATION

1.1 <u>Definitions</u>. As used herein, the capitalized terms below shall have the following meanings. Any of such terms, unless the context otherwise requires, may be used in the singular or plural, depending upon the reference.

"<u>Affiliate</u>" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such Person, and the term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by contract or otherwise.

"Agreement" means this Asset Purchase Agreement, as it may be amended from time to time in accordance with its terms.

"<u>Alcatel Lucent Agreements</u>" means, collectively, the Asset Purchase Agreement dated September 2, 2008 by and among Alcatel Lucent, Alcatel Micro Machining Systems ("<u>AMMS</u>") and Seller, and agreements entered into by Seller with Alcatel Lucent or AMMS or any of their respective Affiliates pursuant thereto or in connection therewith or relating thereto. "Business Day" means any day of the year on which banking institutions in both the State of California and the City of London, England are open to the public for conducting business and are not required or authorized to close.

"Business Records" means organizational documents, business permits, share registers (including with regard to Tegal France the shares official register ("registre des mouvements de titres" and "comptes individuels d'actionnaires")), capitalization tables, financial statements, accounting ledgers, budgets, forecasts, catalogues, sales literature, marketing and advertising materials, bill of materials and kitting lists, technical manuals, engineering drawings, schematics, quality documentation, test and assembly documentation, customer and supplier lists, production records, operating data and plans, research and development notes and reports, employment records, regulatory files, correspondence and any other books, records, files, documents and information, whether maintained in paper, electronic or any other format or medium.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Contract</u>" means any contract, agreement, indenture, note, bond, mortgage, loan, instrument, lease, license, commitment or other arrangement, understanding, undertaking, commitment or obligation, whether written or oral.

"<u>Copyrights</u>" means United States and foreign copyrights, rights under copyrights, and original works of authorship, whether registered or unregistered, together with any registrations and applications for registration thereof.

"Employee Benefit Plan" means an employee compensation or benefit plan, program or arrangement of any kind, and each plan, program or agreement, written or oral, statutory or contractual, that provides for compensation or benefits, including any deferred compensation (including any nonqualified deferred compensation plan within the meaning of §409A of the Code), pension or retirement, workers compensation, medical, life insurance, disability or other welfare plan, educational assistance, loan, expatriate relocation or assistance, unfulfilled relocation obligations, severance, change in control, retention, employment, holiday, cafeteria, vacation, incentive, bonus, stock option, stock purchase, and restricted stock plan, program, or arrangement, and any form of benefit described in any applicable offer letter or employment agreement or engagement under which any employee has any present or future right to benefits.

"Employees" means, collectively, the employees of Seller and Tegal France.

"Encumbrance" means any lien, charge, mortgage, covenant, easement, claim, encumbrance, restriction, security interest, pledge, or title defect, whether arising by Contract or by operation of Law.

"Environmental Law" means any Law as now or hereafter in effect in any way relating to the protection of human health and safety, the environment or natural resources.

"ERISA" means the United States Employee Retirement Income Security Act of 1974, as amended.

"Escrow Agent" means U.S. Bank, N.A.

"Escrow Agreement" means the agreement entered into by and among Seller, Purchaser and the Escrow Agent at Closing in substantially the form attached hereto as Exhibit A.

"Excluded Businesses" means all of Seller's business activities other than the Included Businesses.

"<u>French Subsidiary Shares</u>" means all of the 1,176,556 issued shares of Tegal France, representing one hundred percent (100%) of the securities of, and the ownership and proprietary interests in, Tegal France as of the Closing.

"GAAP" means generally accepted accounting principles in the United States.

"<u>Governmental Authority</u>" means any supranational, international or federal legislative, regulatory, administrative or judicial body, including any court or tribunal, or governmental authority, whether foreign or domestic; each political subdivision thereof; any agency, authority, ministry, bureau, department, commission, instrumentality or officer thereof exercising executive, legislative, judicial, regulatory, taxing or administrative functions, and any quasi governmental or regulatory agency, body or authority, including the European Union, and the European Central Bank.

"IFRS" means International Financial Reporting Standards.

"Included Assets" means the Purchased Assets and the Tegal France Assets.

"Included Contracts" means the Assigned Contracts and the Tegal France Contracts.

"Included Liabilities" means the Assumed Liabilities and the Tegal France Liabilities.

"Included Intellectual Property" means the Purchased Intellectual Property and the Tegal France IP.

"Intellectual Property" means all of the following, whether arising or existing under the laws of the United States or any other jurisdiction: (a) Patents; (b) Marks and the goodwill associated therewith; (c) Copyrights, mask works, proprietary source code, advertising and marketing materials and other works of authorship; (d) uniform resource locator (URL) and internet domain name registrations; (e) rights in databases and data collections (including knowledge databases, customer lists and customer databases), whether registered or unregistered, and any applications for registration therefor; (f) trade secrets and other rights in know-how and confidential or proprietary information (including any inventions, improvements, formulations, discoveries, designs, research, business plans, drawings, notes, technical data, customer data, financial information, pricing and cost information, bills of material or other similar information); (g) rights to bring any demand, action or claim for infringement, including the right to receive all damages, royalties, payments, fees and other amounts related thereto; and (h) other proprietary, industrial, or intellectual property rights now known or hereafter recognized to the full extent that the same may be susceptible to or recognized and/or extended any form of protection by a Governmental Authority.

"Inventory" means finished goods, work-in-process inventory, raw materials, spare and replacement parts, and repair and maintenance materials.

Treasury.

"IRS" means the United States Internal Revenue Service and, to the extent relevant, the United States Department of

"<u>Knowledge of Seller</u>" means the knowledge of Thomas R. Mika, Christine T. Hergenrother or Nicolas Launay. An individual has "knowledge" of a fact for purposes of this definition when the individual has actual knowledge of it or of facts which would indicate to a reasonable person the existence thereof.

"Law" means any supra-national, European Union, national, federal, state, regional, local or other law, rule, statute, ordinance, Order, regulation or subordinate legislation or common law or civil code of any Governmental Authority of any competent jurisdiction, whether in the United States or elsewhere.

"<u>Liability</u>" means any liability, debt, responsibility or obligation of any kind (whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, whether or not of the kind required to be reflected in a balance sheet prepared in accordance with GAAP (with respect to Seller Parties) or IFRS (with respect to Tegal France), and whether due or to become due).

"<u>Mark</u>" means any United States or foreign trademark, service mark, trade dress, logo, trade name or corporate name, whether registered or unregistered, together with any registrations, and applications for registration thereof.

"<u>Material Adverse Effect</u>" means any event, effect, circumstance or change that, individually or together with all other events, effects, circumstances or changes, is or would reasonably be expected to be materially adverse to the value or utility of the Included Assets taken as a whole, or to the ability of Seller to consummate the transactions contemplated in this Agreement or the other Transaction Documents taken as a whole; *provided, however*, that "Material Adverse Effect" shall not be deemed to include the impact of any events, effects, circumstances or changes to the extent they result from (a) changes in the general economic conditions in the United States or France that do not disproportionately impact the Included Assets in any material respect; (b) changes generally in the semiconductor industry that do not have a disproportionate effect on the Included Assets taken as a whole in any material respect; (c) the announcement of the acquisition of the Purchased Assets or the pendency of the transactions contemplated hereby; (d) the performance by Seller of its obligations under this Agreement and the other Transaction Documents or otherwise taken with the express written consent of Purchaser; (e) changes in GAAP (with respect to Seller Parties) or IFRS (with respect to Tegal France), or applicable Laws after the Effective Date; (f) acts of God, calamities, acts of war, terrorism or military action or the escalation thereof.

"Order" means any order, injunction, judgment, doctrine, decree, ruling, writ, assessment or arbitration award of a Governmental Authority.

"Other Subsidiaries" means the subsidiaries of Seller other than Tegal France.

"<u>Patents</u>" means United States and foreign patents, patent applications, registered design, registered design applications, disclosures relating thereto (and any patents that issue as a result of those patent applications), any renewals, reissues, reexaminations, extensions, continuations, continuations-in-part, divisions and substitutions relating to any of the patents, and patent applications.

"<u>Permits</u>" means all licenses, permits, franchises, accreditations, certifications, certificates of need, approvals, authorizations, consents or Orders of, or filings with, any Governmental Authority.

"Permitted Encumbrance" means (i) statutory liens for current Taxes, assessments or other governmental charges not yet delinquent or the amount or validity of which is being contested in good faith by appropriate Proceedings (for the avoidance of doubt, Seller shall be responsible for the payment of all such Taxes to the extent provided by Section 6.2); (ii) mechanics', carriers', workers', repairers' and similar liens arising or incurred in the ordinary course of business that are not material to the Included Businesses or the Included Assets or Assumed Liabilities and that are not resulting from a breach, default or violation by any Seller Party of any Contract or instrument for charges not yet delinquent or the amount or validity of which is being contested in good faith by appropriate Proceedings; (iii) zoning, entitlement and other land use and environmental regulations or other requirements by any Governmental Authority provided that such regulations or other requirements have not been violated; (iv) licenses to Purchased Intellectual Property granted by Seller to third parties on or before the Closing Date and in existence as of the Closing Date and as listed on Section 4.7(a) of the Disclosure Letter; and (v) such other imperfections in title, charges, easements, restrictions and encumbrances which do not materially detract from the value of or materially interfere with the present use of any Included Asset subject thereto or affected thereby; provided, that any Liabilities arising out of any contested charges or assessments under clause (i) or contested liens under clause (ii) shall be Excluded Liabilities for which Purchaser shall be indemnified under ARTICLE VII.

"Person" means an individual or an association, corporation, partnership, trust, firm or any other entity, group or organization, including a Governmental Authority.

"<u>Post-Closing Tax Period</u>" means any Tax period beginning after the Closing Date and that portion of any Straddle Period beginning after the Closing Date.

"<u>Pre-Closing Tax Period</u>" means any Tax period ending on or before the Closing Date and that portion of any Straddle Period ending on the Closing Date.

"<u>Proceeding</u>" means any action, suit, litigation, arbitration, expert dispute resolution or determination, hearing, inquiry, audit, examination, investigation or other proceeding threatened in writing, commenced, brought, conducted or heard by or before, or otherwise involving, any court or other Governmental Authority or any arbitrator or arbitration panel, including any civil, criminal, administrative, investigative or appellate proceeding, whether formal or informal.

"Property Taxes" means real property Taxes, personal property Taxes and similar ad valorem obligations.

"Registered Intellectual Property" means Intellectual Property that is the subject of an application, certificate, filing, registration or other document issued by, filed with, or recorded by any Governmental Authority (whether provisional, supplemental, or otherwise), anywhere in the world.

"Seller Parties" means Seller and any Other Subsidiary that owns, leases or has any rights with respect to the Purchased Assets.

"<u>Software</u>" means software, firmware, programs and databases in any form, including web content, machine code, assembly code, source code, executable code, tools, developers kits, utilities, graphical user interfaces, menus, images, icons and forms, and all versions, updates, corrections, enhancements and modifications thereof, all class, data files, structures, header and include files, and all related documentation, developer notes, comments and annotations relating thereto.

"<u>Tangible Personal Property</u>" means machinery, equipment, furniture, fixtures, devices, materials, spare parts, supplies, computers, servers, routers, networking components, office equipment and supplies and other items of tangible personal property.

"<u>Tax</u>" means any tax, levy, impost, fee, duty, assessment or other government charge, including income, estimated income, business, occupation, franchise, property, payroll, personal property, sales, value-added, excess profits, transfer, use, employment, commercial rent, occupancy, franchise or withholding taxes, and any interest, penalties and additions thereto, imposed by or owing to any Governmental Authority.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any election, schedule or attachment thereto and any amendment thereof, required to be filed with a Governmental Authority.

"Taxing Authority" means the IRS and any other Governmental Authority responsible for the administration of any Tax.

"<u>Tegal France</u>" means Tegal France, SAS, a French *société par actions simplifiée* with a share capital of  $\in$ 1,176,556, with registered office located at 4, avenue du pré de Challes – 74940 Annecy le Vieux, France, registered with the Commercial Registry under number 511 899 544 R.C.S. Annecy and a wholly-owned subsidiary of Seller.

"Tegal France Contracts" means the Contracts to which Tegal France is a party.

"Tegal France IP" means the Intellectual Property owned by Tegal France.

"Tegal France Liabilities" means the Liabilities of Tegal France that are not Excluded Liabilities.

"<u>Trademark License Agreement</u>" means the trademark license agreement entered into by and among Seller and Purchaser at Closing in substantially the form attached hereto as <u>Exhibit D.</u>

"Transaction Documents" means this Agreement, the Trademark License Agreement, the Assignment and Assumption Agreement, the Bill of Sale, the Escrow Agreement, the IP Assignments and the Mutual Release.

(a) <u>Other Defined Terms</u>. The following terms are defined in the Sections set forth below:

Term	Section	
Allocation	2.8(a)	
Allocation Schedule	2.8(a)	
Assigned Contracts	2.1(f)	
Assignment and Assumption Agreement	3.2(a)	
Assumed Liabilities	2.3	
Available Cash	2.6(b)(i)	
Balance Sheet	4.6	
Balance Sheet Date	4.6	
Basket	7.4(a)	
Bill of Sale	3.2(a)(ii)	
Business Employees	4.19(a)	
Cash Consideration	2.5(a)	
Cash Deficiency	2.6(e)	
Cash Surplus	2.6(d)	
Closing	3.1	
Closing Date	3.1	
Closing Payment	2.5(c)(i)	
Compact Business	Recital A	
Compete	6.7	
Current Assets	2.6(c)(ii)	
Current Liabilities	2.6(c)(iii)	
Customer Prepayments	2.6(c)(iv)	
Damages	7.2	
Designation Notice	2.1	

Term	Section
Designee	2.1
Disclosure Letter	ARTICLE IV
DRIE Business	Recital A
Effective Date	Introductory paragraph
Escrow Funds	2.5(c)(ii)
Excluded Assets	2.2
Excluded Contracts	2.2(c)
Excluded Liabilities	2.4
Financial Statements	4.6
Included Businesses	Recital A
Included Permits	4.10(b)
Indemnified Party	7.5
Indemnifying Party	7.5
IP Assignments	3.2(a)(vi)
Lower Cap	7.4(b)
Minimum Cash	2.6(c)(v)
Mutual Release	3.2(a)(xi)
Net Working Capital	2.6(c)(vii)
Nonassignable Assets	2.7
Party, Parties	Introductory paragraph
Pluto Business	Recital A
Pre-Closing Statement	2.6(b)
Purchase Price	2.5(a)
Purchase Price Cap	7.4(b)
Purchased Assets	2.1
Purchased Intellectual Property	2.1(c)
Purchaser	Introductory paragraph
Purchaser Indemnified Parties	7.2
Seller	Introductory paragraph
Seller Indemnified Parties	7.3
Straddle Period	6.2(b)
Tax Matter	6.2(g)(i)
Tegal Employee Plan	4.20(a)
Tegal France Assets	2.1(k)
Tegal France Employee	4.19(d)

Term	Section	
Third-Party Intellectual Property	4.7(b)	
Transfer Taxes	6.2(a)	
Transferred Employee	6.5	
Unresolved Claim	7.6(b)	
WARN Obligations	4.19(h)	

1.2 Other Definitional and Interpretive Matters.

apply:

(a)

Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall

(i) <u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which, any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is not a Business Day, the period in question shall end on the next succeeding Business Day.

(ii) <u>Dollars</u>. Any reference in this Agreement to \$ shall mean United States dollars.

(iii) <u>Accounting Terms</u>. Accounting terms used and not otherwise defined in this Agreement with respect to Seller Parties shall have the meanings given to them under GAAP. Accounting terms used and not otherwise defined in this Agreement with respect to Tegal France shall have the meanings given to them under IFRS.

(iv) <u>Exhibits/Schedules</u>. The Exhibits and Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

(v) <u>Gender and Number</u>. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.

(vi) <u>Headings</u>. The division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Article," "Section" or "Schedule" are to the corresponding Article or Section of or Schedule to this Agreement unless otherwise specified.

(vii) <u>Herein</u>. Words such as "herein," "hereinafter," "hereof" and "hereunder" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

(viii) <u>Including</u>. The word "including" or any variation thereof means (unless the context of its usage requires otherwise) "including, but not limited to," and shall not be construed to limit any general statement that it follows to the specific or similar items or matters following it, nor shall any preceding general statement limit the specific or similar items or matters following it.

(b) The Parties have participated jointly in the negotiation and drafting of this Agreement and if any ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the Parties and no presumption or burden of proof shall arise favoring or disfavoring either Party by virtue of the authorship of any provision of this Agreement.

## ARTICLE II

## PURCHASE AND SALE OF ASSETS

2.1 <u>Purchased Assets</u>. Subject to the terms and conditions set forth in this Agreement, Seller shall, and shall cause the Other Subsidiaries to, sell, assign, transfer and deliver to Purchaser or one or more Persons designated by Purchaser pursuant to this Section 2.1 (each, a "<u>Designee</u>"), and Purchaser agrees to, and to cause its Designees to, purchase, acquire and accept from the Seller Parties, all of the following, free and clear of any Encumbrance other than Permitted Encumbrances (the "<u>Purchased Assets</u>"):

(a) <u>Tegal France</u>. The French Subsidiary Shares, together with all rights attaching to the French Subsidiary Shares at Closing, including the right to receive all the dividends or distributions of any kind declared or paid after Closing Date;

(b) <u>Product Lines</u>. All product lines of the Included Businesses, including those described in Schedule 2.1(b), and any product improvements and future product developments and plans relating thereto; <u>provided</u>, <u>however</u>, that any such product improvements and business plans are provided without any warranty as to the quality or utility thereof;

(c) <u>Intellectual Property</u>. All Intellectual Property owned by Seller and the Other Subsidiaries that relates primarily to, or is used or held for use primarily in, the Included Businesses, including the Patents and Marks listed in Section 4.7(e) of the Disclosure Letter (the "<u>Purchased Intellectual Property</u>");

(d) <u>Inventory</u>. All Inventory of the Seller and the Other Subsidiaries used or held for use in the Included Businesses, wherever located;

(e) <u>Tangible Personal Property</u>. All Tangible Personal Property of the Seller and the Other Subsidiaries (other than fixtures, leasehold improvements, computers, servers, routers, networking equipment, phones and other office machines and equipment and supplies) that is located in the Petaluma Facility and is used or held for use primarily in the Included Businesses;

(f) <u>Assigned Contracts</u>. All rights of the Seller and the Other Subsidiaries under the Contracts listed in Schedule 2.1(f) (collectively, the "<u>Assigned Contracts</u>"), together with any claims, deposits, prepayments, refunds, causes of action, choses in action, rights of recovery, rights of set off, and rights of recoupment of Seller and the Other Subsidiaries arising under any of the foregoing, *provided* that Seller shall retain the right to receive any amounts payable under those Assigned Contracts listed in Schedule 6.6(b)(ii);

(g) <u>Records</u>. All Business Records that relate primarily to Tegal France or the Included Businesses, wherever located, but excluding the Business Records described in Section 2.2(i); provided that Seller may retain a copy of such Business Records;

(h) <u>Permits</u>. All Permits used primarily in the Included Businesses;

(i) <u>Alcatel Lucent Assets</u>. All assets, properties and rights acquired from Alcatel or AMMS pursuant to the Alcatel Lucent Agreements, except for (i) those assets, properties and rights that have been or are disposed of in the ordinary course of business prior to the Closing Date and (ii) any rights of Seller under Excluded Contracts;

(j) <u>Goodwill</u>. All goodwill and similar intangible assets of the Seller and the Other Subsidiaries associated primarily with the Included Businesses; and

(k) <u>General</u>. All property, assets and rights of the Seller and the Other Subsidiaries of every kind, character and description, tangible or intangible, contingent or absolute, wherever located, which are used or held for use primarily in, or that relate primarily to, the Included Businesses, whether or not similar to the items listed in Sections 2.1(a) through (k).

The Parties agree and acknowledge that Tegal France shall retain all of the assets, properties and rights it holds at the time of the Closing (the "<u>Tegal France Assets</u>") and that Purchaser shall automatically acquire from Seller at Closing, by virtue of the transfer of the French Subsidiary Shares to Purchaser, all of Seller's indirect interest in, and control over, the Tegal France Assets. Purchaser may designate one or more of its wholly-owned subsidiaries to acquire and accept any or all of the Purchased Assets from Seller pursuant to this Section 2.1 or to assume any or all of the Assumed Liabilities from Seller pursuant to Section 2.3 below, or both, by delivering to Seller written notice of such designation (a "Designation Notice"), duly executed by Purchaser and acknowledged by each such Designee, no later than the Closing.

2.2 <u>Excluded Assets</u>. Notwithstanding anything to the contrary in this Agreement, Seller shall not, and shall not cause or allow the Other Subsidiaries to, sell, assign, transfer or convey to Purchaser or any Designee any assets or properties of the Seller Parties other than the Included Assets (collectively, the "<u>Excluded Assets</u>"), including the following:

(a) <u>Equity Interests</u>. Any equity securities or other ownership interests of any Seller Party in any subsidiary thereof or in any other Person other than the French Subsidiary Shares;

(b) <u>Product Lines</u>. All product lines of the Excluded Businesses, and any product improvements and future business

(c) <u>Accounts Receivable</u>. All accounts receivable as of the Closing Date of the Seller and the Other Subsidiaries relating to the Included Businesses;

(d) <u>Excluded Contracts</u>. All of the Contracts listed in Schedule 2.2(c) and any other Contracts to which a Seller Party is party that relate primarily to the Excluded Businesses and are not listed in Schedule 2.1(f) (the "<u>Excluded Contracts</u>");

(e) <u>Intellectual Property</u>. All Intellectual Property owned by Seller and the Other Subsidiaries that relates exclusively to, or is used or held for use exclusively in, the Excluded Businesses;

(f) <u>Government Reimbursements</u>. The portion of any joint development project expense reimbursement received after the Closing from a French Governmental Authority that relates to (i) expenditures paid by Seller or Tegal France prior to the Closing or (ii) expenditures incurred by Seller or Tegal France prior to the Closing that are not Included Liabilities;

(g) <u>Cash</u>. All cash and cash equivalents held by the Seller Parties, subject to Section 2.6;

(h) <u>Tangible Personal Property</u>. Tangible Personal Property consisting of fixtures, leasehold improvements, computers, servers, routers, networking equipment, phones and other office machines and equipment and supplies used exclusively at Seller's facility in Petaluma, California; and all items of Tangible Personal Property that are used or held for use primarily in the Excluded Businesses, wherever located;

(i) <u>Business Records</u>. All minute books, organizational documents, share registers and other books and records that pertain exclusively to the ownership, organization or existence of any Seller Party other than Tegal France; all personnel files and other employment records of the Seller Parties that do not pertain to current or former employees of Tegal France or any other Business Employees; and all other Business Records of the Seller Parties that pertain exclusively to the Excluded Businesses;

(j) <u>Tax Claims and Attributes</u>. All claims of the Seller Parties and Tegal France with respect to income Tax refunds, allowances, credits, deductions, exemptions, net operating loss carryovers, set-offs or other Tax attributes, except such items with respect to Tegal France as are allocated to Purchaser under Section 6.2; and

(k) <u>General</u>. All rights, properties and assets of Seller and its subsidiaries that pertain exclusively to the Excluded Businesses.

2.3 <u>Assumed Liabilities</u>. Notwithstanding anything to the contrary in this Agreement, Seller shall not, and shall not permit any of the Other Subsidiaries to, transfer, convey, assign or deliver, and Purchaser shall not, and shall not cause any of its Designees to, acquire or assume, any Liabilities of the Seller Parties other than the Assumed Liabilities as specifically described in this Agreement. Without limiting the generality of the foregoing, upon the terms and subject to the conditions of this Agreement, Purchaser shall, or shall cause its Designee(s) to, assume, pay, perform and discharge when due only the following Liabilities (collectively, the "<u>Assumed Liabilities</u>"):

Assigned Contracts. All Liabilities under the express terms of the Assigned Contracts arising after the Closing (a) Date, except to the extent the Liability or obligation arises out of any breach, act or omission that occurred on or prior to the Closing Date;

Warranty Obligations. All Liabilities under the express terms of warranties pertaining to the Included Businesses (b)that are described in Section 4.16 of the Disclosure Letter:

Post-Closing Operation. Any Liabilities arising after the Closing from the ownership or operation of the Included (c)Assets or the Included Businesses after the Closing; and

Other Commitments. All Liabilities pursuant to the written terms of open purchase orders, open customer orders for (d) spares and additional customer commitments, in each case to the extent (and only to the extent) expressly set forth on Schedule 2.3(d).

Excluded Liabilities. Notwithstanding anything to the contrary contained in this Agreement, neither Purchaser nor any of its 2.4 Designees shall assume or be liable for or otherwise be obligated to pay, perform or discharge, and Tegal France shall not be liable for or otherwise be obligated to pay, perform or discharge, any Liabilities of any Seller Party of any nature whatsoever, whether accrued or unaccrued, absolute or contingent, known or unknown, and regardless of when asserted, other than the Assumed Liabilities (all such Liabilities of the Seller Parties that neither Purchaser nor any of its Designees are assuming are the "Excluded Liabilities"). For the avoidance of doubt, the Excluded Liabilities include:

(a) All Liabilities of any Seller Party or Tegal France that relate primarily to the Excluded Assets or the Excluded Businesses;

(b) All Liabilities of any Seller Party that arise on or at any time after the Closing Date other than the Assumed Liabilities;

All Liabilities of Seller or any Affiliate of Seller other than Tegal France for Taxes, whether accrued or arising prior (c) to, at or after the Closing Date, including any Liability for Transfer Taxes;

All Liabilities of Tegal France other than Current Liabilities incurred in the ordinary course of business prior to the (d) Closing that are expressly listed in the Pre-Closing Statement, to the extent (and only to the extent) taken into account in the calculation of Net Working Capital. Without limiting the foregoing, the following Liabilities of Tegal France are Excluded Liabilities:

> (i) all long-term liabilities incurred on or before the Closing Date;

(ii) any income Tax obligations of Tegal France for Pre-Closing Tax Periods;

Closing;

(iii) Liabilities that arise from the violation of law or breach of contractual or other obligations prior to the

prior to the Closing;

- (iv) Liabilities arising out of environmental claims relating to events that occurred or conditions that existed
- (v) Liabilities arising out of defect claims with respect to products delivered prior to the Closing; and

(vi) Liabilities arising out of infringement claims relating to the Included Intellectual Property that are based on actions or omissions that occurred prior to the Closing Date;

(e) All Liabilities of the Seller Parties and Tegal France for borrowed money or the deferred payment of purchase price or otherwise with respect to debt obligations;

(f) All Liabilities of the Seller Parties for trade accounts payable;

(g) All Liabilities relating to employees, consultants and directors of the Seller Parties, including with respect to payroll, vacation, sick leave, workers' compensation, unemployment benefits, pension benefits, or any employment, severance, retention or termination agreement;

(h) All claims by equity or debt holders of any of the Seller Parties, including with respect to this Agreement and the transactions contemplated hereby, and all obligations, losses and other liabilities arising therefrom;

(i) All debts, obligations and other Liabilities of the Seller Parties that do not relate to the Included Businesses;

(j) All Liabilities relating to legal fees and expenses and other transactional costs incurred by the Seller Parties and Tegal France with respect to this Agreement and the transactions contemplated hereby; and

(k) All debts, obligations and other Liabilities of the Seller Parties of any nature whatsoever that are not expressly assumed by Purchaser in Section 2.3.

## 2.5 <u>Purchase Price</u>.

(a) <u>Total Consideration</u>. The aggregate consideration (the "<u>Purchase Price</u>") for the Purchased Assets shall be (a) one million six hundred one housand one hundred fifty-nine dollars and twenty-three cents (\$1,601,159.23) in cash, subject to adjustment pursuant to Section 2.6 (the "<u>Cash Consideration</u>"), and (b) the assumption of the Assumed Liabilities.

(b) <u>Taxes</u>. The Purchase Price shall be exclusive of any Transfer Taxes, value added Tax, custom duties or any Taxes applicable to the transfer of any Purchased Assets.

(c) <u>Payments</u>. On the Closing Date:

(i) Purchaser shall deliver one million four hundred one thousand one hundred fifty-nine dollars and twentythree cents (\$1,401,159.23) of the Cash Consideration, subject to adjustment pursuant to Section 2.6 (the "<u>Closing Payment</u>"), to Seller by wire transfer of immediately available funds to an account designated by Seller in writing at least two Business Days prior to the Closing Date; and

(ii) Purchaser shall deliver \$200,000 of the Cash Consideration (as adjusted from time to time in accordance with the Escrow Agreement, the "Escrow Funds") by wire transfer of immediately available funds to an account specified by the Escrow Agent, to be held and disbursed by the Escrow Agent in accordance with the Escrow Agreement.

## 2.6 <u>Tegal France</u>.

(c)

(a) Prior to Closing Date, Seller shall, and shall cause each of its subsidiaries to, take such action prior to the Closing as may be necessary or advisable to ensure that (i) any debts, obligations and other liabilities of Tegal France that are Excluded Liabilities are repaid, performed or otherwise discharged prior to the Closing (except, in the case of any Tax liabilities of Tegal France, to the extent such liabilities are not due and payable, disregarding for this purpose any extension of time for payment); and (ii) Tegal France shall have no Liabilities at Closing other than Current Liabilities that, taken together, do not exceed Current Assets.

(b) Seller has previously delivered to Purchaser a statement (the "<u>Pre-Closing Statement</u>"), certified by the Chief Executive Officer and Chief Financial Officer of Seller, setting forth:

(i) Seller's calculations of (A) Current Assets, (B) Current Liabilities, (C) Net Working Capital, (D) Available Cash, (E) Customer Prepayments, (F) Employee Retention Bonuses, (F) Minimum Cash and (G) the Cash Deficiency or Cash Surplus (if any), in each case as of the Closing Date; and

(ii) a statement that Tegal France has no Liabilities other than Current Liabilities that were incurred in the ordinary course of business.

with IFRS:

For purposes of this Agreement, the following terms shall have the following meanings, determined in accordance

(i) "<u>Available Cash</u>" means cash and cash equivalents of Tegal France;

(ii) "<u>Current Assets</u>" means the sum of the book values of (A) Available Cash, (B) accounts receivable of Tegal France (net of allowances for bad debt), (C) prepaid expenses of Tegal France, and (D) raw materials purchased by Seller in order to fill open orders under the Assigned Contracts listed in Schedule 6.6(b)(ii);

(iii) "<u>Current Liabilities</u>" means the sum of the book values of (A) accounts payable of Tegal France, (B) accrued Liabilities of Tegal France (other than Excluded Liabilities), (C) Customer Prepayments, (D) Employee Retention Bonuses, and (E) any other accrued Liabilities of the Seller Parties that are Assumed Liabilities (including accrued amounts for installation and warranty obligations under Assigned Contracts);

(iv) "<u>Customer Prepayments</u>" means the total book value of customer prepayments and deposits with respect to those Assigned Contracts listed in Schedule 6.6(b)(ii);

(v) "<u>Employee Retention Bonuses</u>" means the aggregate amount of bonuses payable to Tegal France Employees who may be entitled thereto under the terms of the retention bonus program described in Section 4.19(c) of the Disclosure Letter;

(vi) "<u>Minimum Cash</u>" means the minimum amount of cash and cash equivalents necessary for both (A) Tegal France to have Net Working Capital of not less than \$1.00 and (B) Tegal France's cash and cash equivalents to not be less than the book value of Customer Prepayments at Closing; and

Liabilities.

(vii)

"Net Working Capital" means the difference in the book value of Current Assets, minus Current

(d) If the amount of Available Cash exceeds the Minimum Cash (in each case, as calculated by Seller in the Pre-Closing Statement), the Cash Consideration shall be increased by an amount no greater than the difference between Available Cash and Minimum Cash (the "<u>Cash Surplus</u>")..

(e) If the amount of Minimum Cash exceeds the Available Cash (in each case, as calculated by Seller in the Pre-Closing Statement) the Cash Consideration and Closing Payment payable by Purchaser pursuant to Section 2.5 shall be reduced by the amount of the difference between Minimum Cash and Available Cash (the "<u>Cash Deficiency</u>").

(f) In the event Purchaser determines that the actual Cash Surplus or Cash Deficiency as of the close of business on the Closing Date was different from that stated in the Pre-Closing Statement and notifies Seller of such discrepancy within 30 days after the Closing, a corresponding adjustment payment shall be made by Seller or Purchaser, as the case may be. Absent manifest error, Purchaser's calculation of actual Cash Surplus or Cash Deficiency shall be binding on the Parties.

2.7Nonassignable Assets. Nothing in this Agreement nor the consummation of the transactions contemplated hereby shall be construed as an attempt or agreement to assign any Purchased Asset, including any Contract or Permit, which by its terms or by Law is nonassignable without the consent of a third party or a Governmental Authority or is cancelable by a third party in the event of an assignment ("Nonassignable Assets") unless and until such consent shall have been obtained. Seller shall, and shall cause its Affiliates to, use its commercially reasonable efforts to cooperate with Purchaser at its request for a one year period in endeavoring to obtain such consents promptly (at Seller's expense), provided that Seller shall not be responsible for the payment of any fees not required by the terms of a Nonassignable Asset in connection with its assignment. To the extent permitted by applicable Law, in the event consents to the assignment thereof cannot be obtained, Seller and Purchaser shall cooperate in any lawful and commercially reasonable arrangement, as Seller and Purchaser shall agree, under which Purchaser would, to the extent practicable, obtain the economic claims, rights and benefits under such Nonassignable Asset and assume the economic burdens and obligations with respect thereto in accordance with this Agreement, including by subcontracting, sublicensing or subleasing to Purchaser. Seller shall promptly pay over to Purchaser all money or other consideration received by it in respect of all Nonassignable Assets, and Purchaser shall indemnify and promptly pay Seller for all Assumed Liabilities associated with such Nonassignable Asset. As of and from the Closing Date, Seller authorizes Purchaser, to the extent permitted by applicable Law and the terms of the Nonassignable Assets, at Purchaser's expense, to perform all the obligations and receive all the benefits of Seller under the Nonassignable Assets and appoints Purchaser its attorney in fact to act in its name on its behalf with respect thereto.

## 2.8 <u>Allocation of the Purchase Price</u>.

(a) The Parties agree that the amount of Purchase Price allocable to the French Subsidiary Shares is the amount set forth on Schedule 2.8. In addition, the Purchase Price (including the Assumed Liabilities, to the extent properly taken into account under the Code) shall be allocated among the Purchased Assets and the non-competition covenant of Seller under Section 6.7 of this Agreement as set forth in a schedule (the "<u>Allocation Schedule</u>"). The allocation set forth on the Allocation Schedule (the "<u>Allocation</u>") shall be prepared in accordance with Section 1060 of the Code and the Treasury Regulations thereunder, and corresponding provisions of similar state, local or foreign Tax Laws. Purchaser shall deliver the Allocation Schedule to Seller within sixty (60) days after the Closing Date for Seller's review of such Allocation Schedule, and Purchaser shall reflect in such Allocation Schedule any comments or changes proposed by Seller that are reasonably acceptable to Purchaser no later than ten (10) Business Days after receipt thereof.

(b) Purchaser and Seller (which, for purposes of this Section 2.8(b), shall include their respective Affiliates, to the extent applicable) shall file all applicable Tax Returns (including, without limitation, IRS Form 8594) consistent with the Allocation. Neither Purchaser nor Seller shall take any Tax position inconsistent with such Allocation nor agree to any proposed adjustment to the Allocation by any Taxing Authority without first giving the other Party prior written notice; *provided*, *however*, that nothing contained herein shall prevent Purchaser nor Seller from settling any proposed deficiency or adjustment by any Taxing Authority based upon or arising out of the Allocation, and neither Purchaser nor Seller shall be required to litigate before any court any proposed deficiency or adjustment by any Taxing Authority challenging such Allocation. Not later than thirty (30) days prior to the filing of their respective IRS Forms 8594 relating to this transaction, each of Purchaser and Seller shall be adjusted in a manner consistent with the procedures set forth in this Section 2.8(b) and, to the extent necessary, the Parties shall file supplemental IRS Forms 8594 and deliver a copy to the other Party not later than thirty (30) days prior to such filing.

2.9 <u>Bulk Transfer Laws</u>. Purchaser hereby waives compliance by Seller with the requirements and provisions of any "bulk transfer" Laws of any jurisdiction within the United States that may otherwise be applicable with respect to the sale of any or all of the Purchased Assets to Purchaser; provided, however, that Seller agrees (i) to pay and discharge when due or to contest or litigate all claims of creditors which are asserted against Purchaser or the Purchased Assets by reason of such noncompliance, (ii) to indemnify, defend and hold harmless Purchaser from and against any and all such claims in the manner provided in ARTICLE VII and (iii) to take promptly all necessary action to remove any Encumbrance which is placed on the Purchased Assets by reason of such noncompliance.

#### ARTICLE III

#### THE CLOSING

3.1 <u>Closing Date</u>. The consummation of the purchase and sale of the Purchased Assets and the assumption of the Assumed Liabilities provided for hereunder (the "<u>Closing</u>") shall take place at the offices of Purchaser's counsel (or at such other place as the Parties may designate in writing) concurrently with the execution and delivery of this Agreement by Seller and Purchaser (the "<u>Closing Date</u>"). The Closing shall be deemed to be effective as of 11:59 p.m. Pacific Time on the Closing Date.

3.2 <u>Deliveries</u>. At the Closing:

(a) Seller shall deliver to Purchaser:

(i) the share transfer form ("*ordre de mouvement*") for the French Subsidiary Shares in favor of the Purchaser and the related Tax form (*Cerfa* form n°2759), duly executed by Seller, together with such additional documents and instruments, duly and validly executed by Seller, as may be required to vest in Purchaser or its Designee good and marketable title to the French Subsidiary Shares;

(ii) a bill of sale vesting title in the Purchased Assets (except the French Subsidiary Shares for which the vesting title shall result from the share transfer form mentioned above) to Purchaser or its Designee in the form set forth in <u>Exhibit B</u>, duly executed by Seller and each Other Subsidiary that is transferring any Purchased Assets (the "<u>Bill of Sale</u>");

(iii) an assignment and assumption agreement relating to the Assigned Contracts in the form set forth in <u>Exhibit C</u>, duly executed by Seller and each Other Subsidiary that is transferring any Purchased Assets (the "<u>Assignment and Assumption</u> <u>Agreement</u>");

(iv) the Trademark License Agreement, duly executed by Seller;

(v) evidence of the execution and delivery by the appropriate third parties of each of the consents, waivers and approvals described in Schedule 3.2(a)(v);

(vi) written assignments to Purchaser or its Designee (the "<u>IP Assignments</u>") of all rights in the Patents and Marks that are part of the Purchased Intellectual Property, duly executed by Seller and each Other Subsidiary that is transferring any such Registered Intellectual Property;

(vii) the Escrow Agreement, duly executed by Seller;

(viii) a certificate of non-foreign status of Seller pursuant to Treasury Regulations Section 1.1445-2(b);

(ix) a legal opinion of Richards, Layton & Finger, P.A., special Delaware counsel to Tegal, addressed to Purchaser and Seller and in a form acceptable to both Parties;

(x) letter of resignation duly executed by the director (*président*) of Tegal France;

(xi) an instrument, duly executed by Seller and Tegal France, pursuant to which Seller and Tegal France release each other from certain claims of liability, in a form reasonably satisfactory to Purchaser (the "<u>Mutual Release</u>"); and

(xii) such other documents as Purchaser may reasonably request.

(b) Purchaser shall deliver to Seller:

(i) the Tax transfer forms (*formulaire cerfa n*°2759 *DGI*) with respect to the transfer of all the French Subsidiary Shares duly executed by Purchaser or its Designee;

(ii) the wire transfers described in Section 2.5(c);

(iii) the Assignment and Assumption Agreement, duly executed by Purchaser or its Designee;

- (iv) the Trademark License Agreement, duly executed by Purchaser or its Designee;
- (v) the Escrow Agreement, duly executed by Purchaser; and

(vi) such other documents as Seller may reasonably request such as any Tax transfer forms duly executed by the Purchaser which filing is required in connection with the transfer of the Purchased Assets.

3.3 <u>Physical Delivery of Tangible Assets</u>. To the extent that Included Assets are not actually delivered to Purchaser on the Closing Date, Seller shall effect physical delivery of any Tangible Personal Property acquired by Purchaser by providing Purchaser with physical access to the locations of such Included Assets; *provided, however*, that if Purchaser so requests, Seller shall, at Purchaser's risk and expense, arrange for delivery of the Tangible Personal Property to the destination specified by Purchaser in written instructions provided at or promptly following the Closing.

3.4 <u>Withholding Taxes</u>. Except for value added Taxes, Purchaser and Escrow Agent shall be entitled to deduct and withhold from any payment to any Person under the Transaction Documents such amounts as either of them is required to deduct and withhold with respect to the making of any such payment under any provision of Tax Law. To the extent that amounts are so withheld or deducted and paid over to the appropriate Taxing Authority, such withheld amounts shall be treated for all purposes of the Transaction Documents as having been paid to the Person in respect of which or whom such deduction and withholding was made.

#### ARTICLE IV

#### REPRESENTATIONS AND WARRANTIES OF SELLER

Except as expressly set forth in the letter prepared and executed by Seller and delivered to Purchaser simultaneously with the execution hereof (the "Disclosure Letter"), Seller represents and warrants to Purchaser that all of the statements contained in this ARTICLE IV are true and complete as of the Effective Date (or, if made as of a specified date, as of such date). The information disclosed in any section or subsection of the Disclosure Letter shall be deemed to relate to and to qualify only the particular representations or warranties set forth in the corresponding section or subsection of this Agreement and shall not be deemed to relate to or qualify any other representation or warranty provided herein unless the applicability of such disclosure to such other representation or warranty is reasonably apparent on its face. The information disclosed in any section or subsection of the Disclosure Letter shall not be deemed to constitute an acknowledgment that such matter or item is required to be disclosed therein, and the mere inclusion of an item in the Disclosure Letter as an exception to a representation or warranty shall not be deemed an admission that such item represents a material exception or material fact, event or circumstance or that such item has a Material Adverse Effect.

4.1 <u>Organization and Power</u>. Seller is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware. Seller has all requisite corporate power and authority to carry on the Included Businesses as presently conducted.

## 4.2 <u>Authorization; Enforceability</u>.

(a) Seller has all requisite corporate power and authority to execute and deliver this Agreement and each of the other Transaction Documents to be executed by Seller and to consummate the transactions contemplated hereby and thereby and perform its obligations hereunder and thereunder. The execution, delivery and performance of this Agreement and the other Transaction Documents to which Seller is a party have been duly and validly adopted and approved by a unanimous vote of the board of directors of Seller, and no other corporate proceeding on the part of Seller (including any action by the stockholders of Seller) is necessary to authorize the execution, delivery and performance of this Agreement or any of the other Transaction Documents to which Seller is a party.

(b) This Agreement has been duly and validly executed and delivered by Seller and (assuming due authorization, execution and delivery by Purchaser) constitutes the valid and binding obligations of Seller, enforceable against Seller in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity. The other Transaction Documents, when duly and validly executed and delivered by Seller and the other parties thereto, will constitute the valid and binding obligations of Seller, each enforceable against Seller in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.

## 4.3 <u>Subsidiaries</u>.

### (a) <u>Tegal France</u>.

(iv)

(i) Tegal France is a *société par actions simplifiée* duly organized and validly existing under the Laws of France. Tegal France has all requisite power and authority to carry on the Included Businesses as presently conducted by Tegal France. Tegal France is duly qualified, authorized or licensed to do business as a foreign entity in good standing in every jurisdiction in which the conduct of its businesses requires such qualification, authorization or license. Seller has heretofore delivered to Purchaser complete and correct copies of the bylaws (*statuts*) or equivalent organizational documents of Tegal France, as presently in effect. Thomas R. Mika is the President and sole director of Tegal France.

(ii) Tegal France has an issued share capital of €1,176,556. Other than the French Subsidiary Shares, there are no securities issued or outstanding that are convertible into, or exchangeable or exercisable for, any shares or other securities of Tegal France, nor are there any options, warrants or other rights outstanding to acquire any shares or other securities of Tegal France. Tegal France cannot be required, pursuant to an agreement or any undertaking whatsoever, to issue securities of any kind whatsoever in favor of third parties, including its employees. All outstanding shares of Tegal France have been validly issued and are fully paid, and none has been issued in violation of any preemptive right, right of first refusal or similar right under applicable Law, any of the organizational documents of Tegal France or any other agreement, document or instrument to which Tegal France is a party or by which it is bound. All of the outstanding shares of Tegal France were issued in material compliance with applicable securities Laws.

(iii) The French Subsidiary Shares are held of record by, and owned for the exclusive and sole benefit of, Seller, free and clear of any Encumbrance and no shareholders' agreement or agreement in favor of a third party binds the Seller or Tegal France with any preemptive right, right of first refusal, prohibition to sell, promise to sell or other similar right under applicable Law. Upon consummation of the transactions contemplated by this Agreement to occur at Closing, Purchaser will acquire full legal and beneficial ownership of all of the issued and outstanding shares of Tegal France, free and clear of any Encumbrance.

Included Businesses.

Tegal France has not conducted any activity or incurred any Liabilities other than with respect to the

(b) <u>Other Subsidiaries</u>. None of the Other Subsidiaries holds any assets, properties or rights that primarily relate to, or are used or held for use in, the Included Businesses.

## 4.4 Consents and Approvals; No Violations.

(a) None of the execution and delivery by Seller of this Agreement or the other Transaction Documents, the consummation of the transactions contemplated hereby or thereby, or compliance by Seller with any of the provisions hereof or thereof will conflict with, or result in any violation or breach of, or conflict with or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or the loss of a benefit under, or give rise to any obligation of Seller to make any payment under, or to the increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of any Encumbrance upon any of the Purchased Assets under any provision of (i) the organizational documents of Seller or Tegal France, (ii) any Contract or Permit, (iii) any Order applicable to Seller, Tegal France or the Included Businesses or (iv) any applicable Law, excluding such violations, breaches or defaults which do not have and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

(b) No consent, waiver, approval, Permit or authorization of or filing with, or notification to, any Governmental Authority or other Person (including any of Tegal France's employees) is required on the part of Seller or Tegal France in connection with (i) the execution and delivery of this Agreement or the Transaction Documents, the compliance by Seller with any of the provisions hereof and thereof, the consummation of the transactions contemplated hereby and thereby or the taking by Seller of any other action contemplated hereby or thereby, or (ii) the continuing validity and effectiveness immediately following the Closing of any Included Contract or Included Permit, except as may be required as a result of facts or circumstances relating to Purchaser or its assets or properties.

#### 4.5 <u>Title to Properties; Good Title Conveyed</u>.

(a) Seller owns and has good, valid and marketable title to, or a valid leasehold interest in, and, to the Seller's Knowledge, following the consummation of the transactions contemplated by this Agreement Purchaser will own and have good, valid and marketable title to, or a valid leasehold interest in, all the Purchased Assets, free and clear of any Encumbrance other than Permitted Encumbrances. Tegal France has good, valid and marketable title to, or a valid leasehold interest, free and clear of any Encumbrance Assets, free and clear of any Encumbrance other than Permitted Encumbrance other than Permitted Encumbrances.

(b) The documentation contemplated to be delivered to Purchaser at Closing pursuant to Section 3.2(a) will effectively vest in Purchaser good, valid and marketable title to all of the French Subsidiary Shares held by Seller. The Bill of Sale, Assignment and Assumption Agreement and other instruments to be executed and delivered by Seller to Purchaser at the Closing will effectively vest in Purchaser good, valid and marketable title to, and ownership of, all of the other Purchased Assets to be transferred to Purchaser as contemplated by this Agreement free and clear of any Encumbrance (other than Permitted Encumbrances).

(c) The Included Assets constitute all of the properties and assets of Seller and its subsidiaries that are used or held for use primarily in the Included Businesses. No other assets, properties or rights are necessary for the conduct of the Included Businesses as presently conducted by Seller and its subsidiaries. All items of Tangible Personal Property that are Included Assets are in reasonably good condition (ordinary wear and tear excepted). All of the Tangible Personal Property owned by the Seller Parties that is used or held for use primarily in the Included Businesses is located in the Petaluma Facility.

4.6 <u>Financial Statements</u>. Section 4.6 of the Disclosure Letter sets forth for Tegal France: (i) a statement of assets and liabilities (the "<u>Balance Sheet</u>") as of December 31, 2010 (the "<u>Balance Sheet Date</u>") and (ii) statements of income for the three-month and 9-month periods ended December 31, 2010 (collectively, the "<u>Financial Statements</u>"). The Financial Statements have been prepared from, are in accordance with and accurately reflect, the books and records of Seller and its subsidiaries, have been prepared in accordance with IFRS applied on a consistent basis during the periods involved and fairly present the consolidated financial position and the consolidated results of operations (and changes in financial position, if any) of Tegal France as of the times and for the periods referred to therein (subject to normally recurring year-end audit adjustments which are not material either individually or in the aggregate). Tegal France has no Liabilities other than (a) Liabilities specifically reserved for in the Balance Sheet or (b) Current Liabilities incurred in the ordinary course of business since the Balance Sheet Date that do not exceed, in the aggregate, the Current Assets as of the Closing Date.

## 4.7 <u>Intellectual Property</u>.

(a) <u>Ownership</u>. Seller owns the Purchased Intellectual Property free and clear of any Encumbrances other than Permitted Encumbrances. Tegal France owns the Tegal France IP free and clear of any Encumbrances other than Permitted Encumbrances. Except for Third-Party Intellectual Property, the Included Intellectual Property constitutes all of the Intellectual Property necessary to conduct the Included Businesses in the same manner as conducted by Seller and Tegal France immediately prior to the Effective Date. None of the Other Subsidiaries owns any Intellectual Property that is relevant in any material respect to the Included Businesses or otherwise necessary to conduct the Included Businesses in the same manner as conducted by Seller and Tegal France immediately prior to the Effective Date.

(b) <u>Inbound Licenses and Rights</u>. Other than "shrink-wrap" and similar commercially available end-user licenses, Section 4.7(b) of the Disclosure Letter (i) lists all Intellectual Property that any third party has licensed to Seller or Tegal France or otherwise authorized Seller or Tegal France to use, for use primarily in the Included Businesses (the "<u>Third-Party Intellectual Property</u>"), and (ii) indicates any such Third-Party Intellectual Property that requires the consent of any other Person (A) in connection with a change of control of Tegal France or (B) in order to be assumed, sold, transferred and assigned to Purchaser as contemplated by this Agreement or otherwise in connection with the consummation of the transactions contemplated hereby and by the other Transaction Documents. Neither Seller nor Tegal France has materially breached any of the Contracts governing the Third-Party Intellectual Property, and, to the Knowledge of Seller, no other party to those Contracts has breached those Contracts.

(c) <u>No Restrictions</u>. Neither Seller nor Tegal France has any obligation to pay royalties, license fees or any other consideration to any Person to use, exploit, practice, sell or dispose of the Purchased Intellectual Property. To the Knowledge of Seller, there is no Proceeding or Order that prohibits or restricts, or seeks to prohibit or restrict, any use of the Included Intellectual Property by Seller or Tegal France. To the Knowledge of Seller, no Person has any rights in the Included Intellectual Property that could cause any reversion or renewal of rights in favor of that Person or termination of the Seller's or Tegal France's or, immediately after the Closing, Purchaser's rights in the Purchased Intellectual Property.

(d) <u>Perfection of Ownership Rights</u>. With respect to the Included Intellectual Property, except as set forth on Section 4.7(d) of the Disclosure Letter, since September 16, 2008, Seller has secured valid written assignments, in substantially the form provided to Purchaser, from all consultants and employees of a Seller Party who contributed to the creation or development of Included Intellectual Property of the rights to such contributions that Seller or Tegal France do not already own by operation of Law.

(e) <u>Registered Intellectual Property</u>. Section 4.7(e) of the Disclosure Letter separately lists all Registered Intellectual Property included in the Included Intellectual Property. To the Knowledge of Seller, all necessary registration, maintenance, renewal, and annuity fees and Taxes owed with respect to the material Registered Intellectual Property that is Included Intellectual Property on or before the Closing Date have been paid, and all necessary documents have been filed in connection with the material Registered Intellectual Property that is Included Intellectual Property. There are no adverse actions or proceedings pending or, to the Knowledge of Seller, threatened by or before any Governmental Authority relating to the registrations or applications for such Included Intellectual Property.

(f) <u>Validity</u>. To the Knowledge of Seller, no opposition Proceedings have been commenced related to any item of Registered Intellectual Property included in the Included Intellectual Property in any jurisdictions where such procedures are available.

(g) <u>Outbound Licenses and Rights</u>. Section 4.7(g) of the Disclosure Letter lists all Contracts under which Seller or Tegal France has licensed or otherwise granted rights in any of the Included Intellectual Property to any Person, other than non-exclusive licenses granted by the Seller Parties in the ordinary course of business pursuant to documentation heretofore disclosed to Purchaser. Section 4.7(g) of the Disclosure Letter also lists separately any of the following related to the Included Intellectual Property: (i) any exclusive rights granted to any third Person, (ii) any source code escrow or other form of delivery or disclosure of any source code to or for the benefit of any Person, and (iii) any other agreements that give other Persons the right to use, market or otherwise exploit or commercialize any of the Included Intellectual Property, other than non-exclusive licenses granted by the Seller Parties in the ordinary course of business.

(h) <u>Indemnity Agreements</u>. Neither Seller nor any subsidiary thereof has agreed, outside the ordinary course of business to indemnify, defend or otherwise hold harmless any other Person with respect to damages resulting or arising from the Included Intellectual Property.

(i) <u>No Violation of the Seller Parties' Rights</u>. To the Knowledge of Seller (i) no Person has infringed or misappropriated any of the Included Intellectual Property in any material respect, and (ii) there is no ongoing infringement or misappropriation of any of the Included Intellectual Property. Since September 16, 2008, neither Seller nor Tegal France has commenced or threatened any Proceeding or claim against any Person for infringement or misappropriation of the Included Intellectual Property.

(j) <u>No Violation of Third-Party Rights</u>. To the Knowledge of Seller, neither the conduct of the Included Businesses nor the creation, use, license or other transfer of the Included Intellectual Property by Seller infringe or misappropriate any other Person's Intellectual Property rights. Neither Seller nor any of its subsidiaries has received written notice of any pending or threatened Proceeding or claim in which any Person alleges that Seller or a subsidiary of Seller, in any way related to the Included Businesses or the Included Intellectual Property, has violated any Person's Intellectual Property rights. There are no pending Proceedings or claims between Seller or any of its subsidiaries and any other Person relating to the Included Intellectual Property.

(k) <u>Confidentiality</u>. Seller and Tegal France have taken commercially reasonable steps to protect and preserve trade secrets and other confidential information included in the Included Intellectual Property.

(1) <u>No Special Adverse Circumstances</u>. To the Knowledge of Seller, the Software that is included in the Included Intellectual Property does not contain any source or object code or other Intellectual Property that is not wholly-owned by Seller or Tegal France. Neither Seller nor Tegal France is a member of, nor is either obligated to license or disclose any Intellectual Property to, any official standards setting organization or to any such organization's members. Seller and Tegal France have not used any Software of the type commonly referred to as "freeware" or "shareware" in a manner that requires the Included Intellectual Property to be (i) disclosed or distributed in source code form, (ii) licensed for the purpose of making derivative works or (iii) redistributable at no charge.

4.8 <u>Material Contracts</u>. Section 4.8 of the Disclosure Letter sets forth a complete and accurate list of all Contracts to which Seller or Tegal France is a party and that relate primarily to the Included Assets or the Included Businesses other than Contracts that are not material, individually or in the aggregate, to the operation of the Included Businesses as presently conducted. Each of the Included Contracts is in full force and effect and is the legal, valid and binding obligation of Seller or Tegal France, as applicable, and, to the Knowledge of Seller, of the other parties thereto, enforceable against each of them in accordance with its terms, except to the extent enforcement may be limited by any applicable bankruptcy, reorganization, insolvency, moratorium or similar applicable Law affecting creditors' rights generally and principles governing the availability of equitable remedies. Neither Seller nor Tegal France is in default under any Included Contract, nor, to the Knowledge of Seller, is any other party to any Included Contract in breach of or default thereunder, and to the Knowledge of Seller no event has occurred that with the lapse of time or the giving of notice or both would constitute a breach or default by Seller or Tegal France or any other party thereunder. No party to any of the Included Contracts has exercised any termination rights with respect thereto, and no such party has given written notice, or to the Knowledge of Seller oral notice, of its intent to exercise such rights with respect to any Included Contract, other than in the ordinary course of business. Seller has made available to Purchaser true, correct and complete copies of all of the Included Contracts, together with all amendments, modifications and supplements thereto.

4.9 <u>Litigation</u>. There is no Proceeding with respect to the Included Assets, the Included Businesses or the transactions contemplated hereby pending or, to the Knowledge of Seller, threatened against Seller or Tegal France (or to the Knowledge of Seller, pending or threatened against any of the officers or directors of Seller or Tegal France in their capacities as such), or to which Seller or Tegal France is otherwise a party, before any Governmental Authority. Neither Seller nor Tegal France is in violation of, or subject to, any Order regarding the Included Assets, the Included Businesses or the transactions contemplated hereby. Seller is not presently engaged in any legal action related to the Included Assets, the Included Businesses or the transactions contemplated hereby to recover monies due it or for damages sustained by it.

## 4.10 Compliance with Laws; Permits.

(a) Seller and Tegal France are in compliance in all material respects with all Laws applicable to the Included Businesses. Neither Seller nor Tegal France has received any written notice from any Governmental Authority or been charged with the violation of any Laws. To the Knowledge of Seller, neither Seller nor Tegal France is under investigation by any Governmental Authority with respect to the violation of any Laws applicable to the Included Businesses.

(b) Section 4.10(b) of the Disclosure Letter contains a list of all Permits which are required for the operation of the Included Businesses as presently conducted and any other Permits granted to Seller with respect to the Included Businesses or to Tegal France ("Included Permits"). Neither Seller nor Tegal France is in default or violation, and, to Seller's Knowledge, no event has occurred which, with notice or the lapse of time or both, would constitute a default or violation, of any term, condition or provision of any Included Permit. All material fees required to be paid in connection with the Included Permits have been paid. There are no Proceedings pending or, to the Knowledge of Seller, threatened, relating to the suspension, revocation or modification of any of the Included Permits.

4.11 <u>Real Property</u>. Neither Seller nor Tegal France owns any real property. Section 4.11 of the Disclosure Letter sets forth a complete list of all real property and interests in real property leased, subleased, licensed or used or occupied by Tegal France as lessee or sublessor, including a description of each such real property lease, sublease or other agreement. Tegal France has a valid, binding and enforceable leasehold interest under each lease described in Section 4.11 of the Disclosure Letter. Seller has made available to Purchaser true, correct and complete copies of all such leases, together with all amendments, modifications or supplements, if any, thereto. Tegal France has complied in all material respects with all norms and standards in respect of the construction, occupation and use of the premises and land covered by such leases.
# 4.12 <u>Taxes; Tax Returns</u>.

Tegal France is, and at all times has been, an association taxable as a corporation for U.S. federal income tax (a) purposes. All material Tax Returns required to be filed by Tegal France have been timely filed with the appropriate Taxing Authorities and the information provided on such Tax Returns is complete and accurate in all material respects. Tegal France has not requested or been granted any extension of time within which to file a material Tax Return that has not yet been filed. No claim has been received by Tegal France or any Seller Party that Tegal France is or may be subject to any material Tax for which Tegal France does not file Tax Returns. All Taxes due and payable from or with respect to Tegal France, whether or not shown or required to be shown on a Tax Return, have been paid in full. Tegal France has timely withheld and paid all Taxes required to be withheld and paid with respect to any employee, independent contractor, shareholder or other Person. No deficiency in respect of any material Tax has been proposed or assessed in writing against Tegal France, and Tegal France has not executed with current effect any waiver of or extended the statute of limitations on the audit of any material Tax Return or the assessment or collection of any material Tax. No action, audit, proceeding, or investigation with respect to Taxes of Tegal France is pending or, to the Knowledge of Seller, threatened. Tegal France is not a party to or bound by any Tax allocation, indemnity or sharing agreement. Tegal France has not been a member of an affiliated group filing a consolidated United States federal income Tax Return or a tax consolidation scheme applicable to groups of companies provided for in Article 223A et seq. of the French General Tax Code, and Tegal France has no Liability for the material Taxes of any Person under any provision relating to consolidated, combined or unitary Tax filings, as a transferee or successor, by contract, or otherwise. Tegal France has not been a party to any transaction that was intended to be governed, in whole or in part, by Section 355 or 367 of the Code. Tegal France does not use a favorable or exceptional tax system the continuation of which depends on undertakings issued by Tegal France that must be complied with subsequent to the Closing Date.

(b) No action, audit, proceeding, or investigation with respect to Taxes of any Seller Party is pending or, to the Knowledge of Seller, threatened.

(c) Seller is not a "foreign person" within the meaning of Section 1445(f)(3) of the Code.

(d) None of the Purchased Assets is required to be treated as (i) owned by any other Person pursuant to the provisions of Section 168(f)(8) of the Internal Revenue Code of 1954, as amended and in effect immediately before the enactment of the Tax Reform Act of 1986, or (ii) Tax-exempt use property within the meaning of Section 168(h)(1) of the Code. None of the Purchased Assets directly or indirectly secures any debt the interest on which is Tax exempt under Section 103(a) of the Code.

4.13 <u>Books and Records</u>. The Business Records of Seller and its subsidiaries relating to the Included Businesses or to Tegal France are complete and correct in all material respects. True and complete copies of all minute books and all stock record books of Tegal France have heretofore been delivered to Purchaser. To the Knowledge of Seller, Tegal France has maintained substantially complete and accurate books and records of all its material company proceedings consistent with applicable Law.

4.14 <u>Accounts Receivable</u>. Section 4.14 of the Disclosure Letter sets forth a current list and aging schedule of all accounts receivable of the Included Businesses. All accounts receivable of the Included Businesses, whether or not reflected in Section 4.14 of the Disclosure Letter, represent sales actually made in the ordinary course of business.

4.15 <u>Inventory</u>. Section 4.15 of the Disclosure Letter sets forth a current list of all Inventory of the Included Businesses. All of the Inventory of the Seller Parties (to the extent related to the Included Businesses), and all of the Inventory of Tegal France, whether reflected in Section 4.15 of the Disclosure Letter, consist of a quality and quantity usable and salable in the ordinary course of business, except for items of obsolete materials and materials of below-standard quality, all of which items have been written off or written down on the Balance Sheet to fair market value or for which adequate reserves have been provided therein. All Inventory not written off has been priced at the lower of average cost or market. All work in process and finished goods inventory included in the Purchased Assets or owned by Tegal France is free of any defect or other deficiency.

4.16 <u>Warranty Claims</u>. Section 4.16 of the Disclosure Letter sets forth (a) Seller's customary forms of guaranty, warranty, right of return, right of credit or other indemnity related to the Included Businesses that legally binds Seller or Tegal France in connection with any licenses, goods or services sold by them, true and correct copies of each of which have been delivered to Purchaser, (b) a list of all sales transactions with respect to which there are outstanding warranties and (c) a summary of any non-standard terms included in outstanding warranties. Since September 16, 2008, neither Seller nor any of its subsidiaries has received any written notice of any outstanding or unpaid material warranty or other claims of any kind related to the Included Businesses against such Person.

4.17 <u>Product Liability</u>. Neither Seller nor Tegal France has any material Liability for, and, to Seller's Knowledge, no event has occurred that is likely to result in a material Liability or Proceeding arising out of, any injury to individuals or property as a result of the ownership, possession or use of any product or service of the Included Businesses that is manufactured, sold, leased or delivered by Seller or Tegal France.

4.18 Insurance. Section 4.18 of the Disclosure Letter lists all policies of insurance currently in effect for Tegal France. Tegal France is sufficiently and validly insured with solvent insurers against the risks that are normally insured for, in particular its civil liability, in reasonably sufficient amounts in respect of the risks incurred, and Tegal France has not undertaken any activity other than the activities insured by these policies and those that they have replaced. To the Knowledge of Seller, there is no threatened termination of any such policies or arrangements. There are currently no claims pending against Seller or Tegal France under any insurance policies currently in effect and covering the Included Assets, the Included Businesses or any Business Employees, and all premiums due and payable with respect to the policies maintained by Seller or Tegal France have been paid to date.

## 4.19 Employees.

(a) Set forth in Section 4.19(a) of the Disclosure Letter is a complete listing of the names, job titles and compensation or other remuneration of (i) all employees of Seller who render services in the conduct of the Included Businesses on a regular basis and (ii) all employees of Tegal France (collectively, the "<u>Business Employees</u>"). Tegal France has not offered for sale or previously told any present or past employee of Tegal France that they would own or participate in any way whatsoever in any component of equity or other form of ownership of Tegal France, except as provided by French Law.

(b) Seller and Tegal France have complied in all material respects with applicable Laws related to the employment of Business Employees, including, as applicable, provisions related to wages, hours, leaves of absence, equal opportunity, occupational health and safety, workers' compensation, severance, employee handbooks or manuals, collective bargaining, dismissals, employee loans, employees representation and the payment of social security and other taxes. Neither Seller nor Tegal France has any material Liability under any applicable Law related to the employment of any Business Employees.

(c) Except as set forth in Section 4.19(a) of the Disclosure Letter, no Business Employee will be entitled, as a result of the transactions contemplated by this Agreement, to: (i) receive any payment, reward or benefit of any kind upon or following the Closing; or (ii) receive any enhancement in or improvement to his remuneration, benefits or terms and conditions of employment. Section 4.19(c) of the Disclosure Letter sets forth the terms of any Employee Retention Bonus payable to each Business Employee.

(d) Tegal France has in relation to its employees (the "<u>Tegal France Employees</u>") and former employees: (i) complied with all obligations, awards, orders and recommendations imposed on them or made by or under statute, statutory instrument, European Community or other Law, common law, contract, collective agreement, terms and conditions of employment, staff handbook, and company policy; and (ii) maintained (and continue to maintain) records which are accurate, up-to-date and complete in all material respects regarding the service of such persons including but not limited to adequate and suitable personnel records, records of working time, terms and conditions of employment, payments of statutory or other sick pay, statutory or other maternity pay, disciplinary, grievance and health and safety matters, income tax and social security contributions and termination of employment.

(e) Except as set forth in Section 4.19(c) of the Disclosure Letter, Tegal France has not granted any right to its turnover or its income to its current or former officers, directors or employees. No agreement has been concluded by Tegal France or is binding on it that grants to any Person benefits exceeding those provided for by Law, any applicable collective bargaining agreements or collective labour agreements (in each case as described in Section 4.19(e) of the Disclosure Letter) and standard practice. Except as set forth in Section 4.19(e) of the Disclosure Letter, none of the current employees of Tegal France have been dismissed or have resigned. Tegal France does not have any Liabilities relating to pensions, supplementary pensions, remuneration or allocations due to retirement or other similar benefits to employees, officers and directors of Tegal France, exceeding those provided for by Law, any applicable collective bargaining agreements or collective labour agreements (in each case as described in Section 4.19(e) of the Disclosure Letter), whether or not required to be reported in the Financial Statements.

(f) With the exception of income, income-related, social-charges related and training-related Taxes in respect of the payment period current at the Effective Date, Tegal France does not have any outstanding or undischarged liability to pay to any Governmental Authority in any jurisdiction any material contribution, Tax or other levy arising in connection with the employment or engagement of personnel by Tegal France.

Tegal France is not a party to any collective bargaining agreement or other labor union contract applicable to Tegal (g) France Employees and Seller does not know of any activities or proceedings of any labor union to organize any Tegal France Employees. Notwithstanding the aforementioned, it is mentioned for the avoidance of doubt, that Tegal France is subject to collective bargaining agreement(s) or other labor union contract(s) without being a party thereto, in particular due to ministerial decision(s) (arrêté(s) d'extension) as described in Section 4.19(g) of the Disclosure Letter. To the Knowledge of Seller, Tegal France is in compliance with all applicable Laws relating to employment and employment practices, wages, hours, and terms and conditions of employment including duration of fixed term employment contracts. There is no unfair labor practice charge or complaint pending against Tegal France before any Governmental Authority, nor is there any labor strike, material slowdown or material work stoppage or lockout actually pending or, to Seller's Knowledge. threatened against or affecting Tegal France. Tegal France has not experienced any strike, material slow down or material work stoppage, lockout or other collective labor action by or with respect to Tegal France Employees since Seller's purchase of Tegal France pursuant to the Alcatel Lucent Agreements. There is no representation, claim or petition pending before any Governmental Authority, and no claim concerning representation exists relating to the Tegal France Employees. There are no charges with respect to or relating to Tegal France pending before any Governmental Authority responsible for the prevention of unlawful employment practices; and neither Seller nor Tegal France has received notice from any Governmental Authority responsible for the enforcement of labor or employment laws of an intention to conduct an investigation of Tegal France or that such an investigation is in progress.

(h) The transactions contemplated by this Agreement will not result in a "plant closing" or "mass layoff" within the meaning of the Worker Adjustment and Retraining Notification Act, §29 U.S.C. 2101 et seq., or under any similar Law (separately and collectively the "<u>WARN Obligations</u>"), and Purchaser's failure to hire or extend employment to any Employees in connection with the transactions contemplated by this Agreement will not result in a "plant closing" or "mass layoff" within the meaning of the WARN Obligations.

# 4.20 Employee Benefits.

(a) Seller has delivered or made available to Purchaser an accurate and complete list of all Employee Benefit Plans maintained by Seller or Tegal France for the benefit of Tegal France Employees (each, a "<u>Tegal Employee Plan</u>"). Each Tegal Employee Plan applicable to any Business Employee is in compliance in all material respects with all applicable requirements of ERISA, the Code, and other applicable Law and has been administered in material accordance with its terms and such applicable Law. There are no pending or, to the Knowledge of Seller, threatened, Proceedings, including claims or litigation, audits or inquiries from any Governmental Authority, with respect to any Tegal Employee Plan, other than ordinary and usual claims for benefits by participants and beneficiaries. With the exception of any Tegal Employee Plan and collective bargaining agreement applicable to it, and/or unless listed in Section 4.19(g) of the Disclosure Letter, Tegal France is not party to or bound by any other collective labor agreements including company agreements, group agreements, deferred salary agreements, retirement, pension and mutual obligations that are not mandatory, and pensions plans, and optional and statutory profit sharing agreements.

(b) The consummation of the transactions described in this Agreement will not (i) accelerate the time of payment or vesting or trigger any payment, forfeiture or funding (through a trust or otherwise) of compensation or benefits under, or materially increase the amount payable or create any other material obligation pursuant to, any Tegal Employee Plan or (ii) result in payments which would not be deductible under Section 280G of the Code.

4.21 <u>Environmental Matters</u>. The operations of Seller and its subsidiaries with respect to the Included Businesses are and have been in compliance in all material respects with all applicable Environmental Laws (including with respect to the retention of and compliance with any Permits required thereunder) and no claim has been made or is pending or, to the Knowledge of Seller, threatened, against Seller or Tegal France, and no notice or other information has been received by Seller or Tegal France alleging, with respect to the Included Businesses, that Seller or Tegal France may be in violation of or have any liability or potential liability under any Environmental Law. Seller has delivered or made available to Purchaser complete and accurate copies of all environmental reports, investigations and audits possessed or controlled by Seller that were obtained from, or conducted by or on behalf of such Seller Party, any Governmental Authority or any Person during the past three years and relating to premises currently or previously owned, leased, operated or controlled by Seller or Tegal France. To the Knowledge of Seller, no premises currently or formerly owned or operated by Seller or Tegal France is materially contaminated by any substance the cost of whose remediation under applicable Environmental Law would, if such remediation were required, constitute a Material Adverse Effect.

4.22 <u>Changes Since Balance Sheet Date</u>. Since the Balance Sheet Date, Seller and Tegal France have conducted the Included Businesses only in the ordinary course of business, and neither Seller nor Tegal France has:

(a) sold, disposed of or otherwise transferred or assigned to any Person any of the Purchased Assets except for Inventory and Tangible Personal Property sold in the ordinary course of business;

(b) (i) hired or offered employment to any individual as a Business Employee, (ii) terminated the employment of any Business Employee (nor has any Business Employee terminated his or her employment with the Company), (iii) granted any Business Employee any increase in compensation, benefits, severance or termination pay, (iv) established, adopted, entered into or amended in any material respect any Employee Benefit Plan for Tegal France other than as required by Law or (v) taken any action to accelerate any payments, rights or benefits under any Tegal Employee Plan;

(c) terminated, rescinded, modified, amended or otherwise altered or changed any of the terms or provisions of any material Included Contract, or reduced, discounted, waived or foregone any material payment or right thereunder, or agreed to any compromise or settlement with respect thereto;

(d) entered into any Contract to do any of the foregoing.

4.23 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker, finder or financial advisor for Seller in connection with the transactions contemplated by this Agreement and no Person is or will be entitled to any fee or commission or like payment in respect thereof.

#### ARTICLE V

## REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser represents and warrants to Seller that all of the statements contained in this ARTICLE V are true and complete as of the Effective Date (or, if made as of a specified date, as of such date).

5.1 <u>Organization</u>. Purchaser is a limited company duly organized, validly existing and in good standing under the laws of England and Wales.

5.2 <u>Authorization; Enforceability</u>. Purchaser has all requisite power and authority to execute and deliver this Agreement and each of the Transaction Documents to be executed by Purchaser and to consummate the transactions contemplated hereby and thereby and perform its obligations hereunder and thereunder. The execution, delivery and performance of this Agreement and the other Transaction Documents to which Purchaser is a party have been duly and validly adopted and approved by a vote of the board of directors of Purchaser, and no other corporate proceeding on the part of Purchaser is necessary to authorize the execution, delivery and performance of this Agreement has been duly and validly executed and delivered by Purchaser and (assuming due authorization, execution and delivery by Seller) constitutes the valid and binding obligations of Purchaser, enforceable against Purchaser in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity. The other parties thereto, will constitute the valid and binding obligations of Purchaser, each enforceable against Purchaser in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.

#### 5.3 Consents and Approvals; No Violations.

(a) None of the execution and delivery by Purchaser of this Agreement or the Transaction Documents to which Purchaser is a party, the consummation of the transactions contemplated hereby or thereby, or compliance by Purchaser with any of the provisions hereof or thereof will conflict with, or result in any violation or breach of, or conflict with or default (with or without notice or lapse of time, or both) under any provision of (i) the Articles of Association or other organizational documents of Purchaser, (ii) any Order applicable to Purchaser or (iii) any applicable Law, excluding from the foregoing such violations, breaches or defaults which would not, individually or in the aggregate, have a material adverse effect on the ability of Purchaser to consummate the transactions contemplated by this Agreement or the other Transaction Documents to which Purchaser is a party.

(b) No consent, waiver, approval, Permit or authorization of or filing with, or notification to, any Person or Governmental Authority is required on the part of Purchaser in connection with the execution and delivery of this Agreement or the Transaction Documents to which Purchaser is a party, the compliance by Purchaser with any of the provisions hereof and thereof, the consummation of the transactions contemplated hereby and thereby or the taking by Purchaser of any other action contemplated hereby or thereby, except as would not, individually or in the aggregate, have a material adverse effect on the ability of Purchaser to consummate the transactions contemplated by this Agreement or the Transaction Documents to which Purchaser is a party.

5.4 <u>Litigation</u>. There are no Proceedings pending or, to the knowledge of Purchaser, threatened that seek to prohibit or restrain the ability of Purchaser to enter into this Agreement or the Transaction Documents to which Purchaser is a party or have a material adverse effect on the ability of Purchaser to consummate the transactions contemplated by this Agreement or the Transaction Documents to be delivered by Purchaser at Closing.

5.5 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker, finder or financial advisor for Purchaser in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment in respect thereof.

## ARTICLE VI

# COVENANTS AND AGREEMENTS

6.1 <u>Publicity</u>. Purchaser and Seller shall consult with each other before issuing, and give each other the opportunity to review and comment upon, any press release or other public statements with respect to the transactions contemplated by this Agreement, and shall not issue any such press release or make any such public statement prior to such consultation, except as may be required by applicable Law, court process or the rules and regulations of any national securities exchange or national securities quotation system. The Parties agree that the initial press release to be issued with respect to the transactions contemplated by this Agreement shall be in a form mutually agreed to by the Parties.

## 6.2 <u>Tax Matters</u>.

(a) All foreign, federal, state and local sales, stamp, documentary, recording, value added, registration, conveyance and real estate and other transfer taxes (but not including value added taxes within the meaning of Council Directive 2006/112/EC of November 28, 2006 on the common system of value added tax, which shall be borne by Purchaser), if any, (such taxes, "<u>Transfer Taxes</u>"), due as a result of the purchase, sale or transfer of the Purchased Assets in accordance herewith shall be borne by Seller, whether imposed by law on Seller or Purchaser. Seller shall file all necessary documentation and Tax Returns with respect to such Transfer Taxes. Purchaser shall cooperate in providing Seller upon Seller's request therefor with any information reasonably necessary for the completion of any Tax formalities (including Tax Returns) relating to the transfer of the Purchased Assets, as well as any appropriate duly executed Tax form, resale exemption certifications, proof of payment or any other similar documentation and Seller shall promptly reimburse Purchaser for any Transfer Taxes paid directly by Purchaser.

Subject to Section 6.2(a), with respect to Property Taxes imposed with respect to the Purchased Assets (other than (b) the Tegal France Assets) for a taxable period that includes (but does not end on) the Closing Date (each such period, a "Straddle Period"), Seller shall be liable for the portion of such Taxes attributable to the Pre-Closing Tax Period, and Purchaser shall be liable for the portion of such Taxes attributable to the Post-Closing Tax Period. The portion of any such Straddle Period Tax that is allocable to the Pre-Closing Tax Period will be deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of calendar days of such Straddle Period in the Pre-Closing Tax Period and the denominator of which is the number of calendar days in the entire Straddle Period (the remainder being the amount allocable to the Post-Closing Tax Period). If bills for such Taxes have not been issued as of the Closing Date, and, if the amount of such Taxes for the period including the Closing Date is not then known, the apportionment of such Taxes shall be made at Closing on the basis of the prior period's Taxes. After Closing, upon receipt of bills for the period including the Closing Date, adjustments to the apportionment shall be made by the Parties, so that if either Party paid more than its proper share at the Closing, the other Party shall promptly reimburse such Party for the excess amount paid by them upon presentation of a statement setting forth the amount of reimbursement to which the presenting Party is entitled along with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement. If either Party receives a refund of Taxes for which the other Party is responsible pursuant to this Section 6.2(b), it shall reimburse the other Party its proportionate share of such refund, less any Taxes or other costs imposed with respect to the receipt of such refund.

Subject to Sections 6.2(a) and (b), with respect to Taxes of Tegal France (including Property Taxes imposed with (c) respect to the Tegal France Assets) for a Straddle Period, Seller shall be liable for the portion of such Taxes attributable to the Pre-Closing Tax Period, and Purchaser shall be liable for the portion of such Taxes attributable to the Post-Closing Tax Period. The portion of any such Straddle Period Tax that is allocable to the Pre-Closing Tax Period will be: (i) in the case of Property Taxes, deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of calendar days of such Straddle Period in the Pre-Closing Tax Period and the denominator of which is the number of calendar days in the entire Straddle Period (the remainder being the amount allocable to the Post-Closing Tax Period), and (ii) in the case of all other Taxes, determined as though the taxable year of Tegal France terminated at the close of business on the Closing Date. If bills for such Taxes have not been issued as of the Closing Date, and, if the amount of such Taxes for the period including the Closing Date is not then known, the apportionment of such Taxes shall be made at Closing on the basis of the prior period's Taxes. After Closing, upon receipt of bills for the period including the Closing Date, adjustments to the apportionment shall be made by the Parties, so that if either Party paid more than its proper share at the Closing, the other Party shall promptly reimburse such Party for the excess amount paid by them upon presentation of a statement setting forth the amount of reimbursement to which the presenting Party is entitled along with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement. With respect to other Taxes other than Property Taxes, if either Party receives a refund of Taxes for which the other Party is responsible pursuant to this Section 6.2(c), it shall reimburse the other Party its proportionate share of such refund, less any Taxes or other costs imposed with respect to the receipt of such refund.

(d) Following the Closing, Purchaser shall prepare and timely file, or shall cause to be prepared and timely filed, all Tax Returns in respect of Tegal France or the Purchased Assets that relate to (i) taxable periods ending on or before the Closing Date but that are required to be filed after the Closing Date and (ii) any Straddle Period, and shall make or cause to be made all payments required to be made with respect to such Tax Returns. Seller shall promptly reimburse Purchaser, without duplication, for the portion of such amounts for which Seller is liable pursuant to Sections 6.2(b) and (c) upon presentation of a statement setting forth the amount of reimbursement. Such Tax Returns shall be prepared in a manner consistent with Seller's past practices, except as otherwise required by applicable Law. Purchaser shall, at least twenty (20) Business Days prior to the due date (taking into account any extension) for filing any such income Tax Return, deliver to Seller for Seller's review a draft of such Tax Return, and Purchaser shall reflect in such Tax Return, as filed, any reasonable comments or changes that Seller submits to Purchaser no later than ten (10) Business Days prior to such due date for filing such Tax Return (and, in the case of material Tax Returns other than income Tax Returns, Purchaser shall, to the extent reasonably practicable, deliver to Seller's review a draft of such Tax Returns, Purchaser shall, to the extent reasonably practicable, deliver to Seller 's review a draft of such Tax Returns, Purchaser shall, to the extent reasonably practicable, deliver to Seller's review a draft of such Tax Returns, Purchaser shall, to the extent reasonably practicable, deliver to Seller 's review a draft of such Tax Returns at reasonable time for such review, and shall reflect in such Tax Return, as filed, any reasonable comments or changes that Seller submits to Purchaser within a reasonably sufficient time prior to the due date for filing such Tax Return).

(e) Seller shall be entitled to any refunds (including without limitation any interest paid thereon) or credits of Taxes with respect to Tegal France or the Purchased Assets attributable to any Pre-Closing Tax Period; provided, however, that Purchaser shall have no obligation to apply for or otherwise pursue any such refund or credit. Purchaser shall promptly forward to or reimburse Seller for any such refunds (including without limitation any interest paid thereon) or credits due to Seller after receipt thereof.

(f) Purchaser shall not cause or permit (i) the filing of any amended Tax Return with respect to Tegal France or the Purchased Assets for any Pre-Closing Tax Period or any Straddle Period without Seller's written consent, which consent shall not be unreasonably withheld or delayed, or (ii) the making of any election under Section 338(g) of the Code with respect to the acquisition of the French Subsidiary Shares or any election under Treasury Regulations Section 301.7701-3 with respect to Tegal France effective on or before the Closing Date.

# (g) <u>Tax Contests</u>.

(i) Purchaser and its Subsidiaries and Affiliates (including, after the Closing, Tegal France), on the one hand, and Seller and its Affiliates, on the other hand, shall promptly notify each other upon receipt by such party of written notice of any inquiries, claims, assessments, audits or similar events with respect to Taxes relating to a Pre-Closing Tax Period or a Straddle Period (any such inquiry, claim, assessment, audit or similar event, a "<u>Tax Matter</u>"). Any failure to so notify the other party of any Tax Matter shall not relieve such other party of any liability with respect to such Tax Matters except to the extent such party was actually prejudiced as a result thereof.

(ii) Purchaser shall have control of the conduct of all Tax Matters, including any settlement or compromise thereof, provided, however, that (A) Seller shall have the right to participate in any such Tax Matter at its own expense and (B) Purchaser shall keep Seller reasonably informed of the progress of any Tax Matter and shall not effect any such settlement or compromise with respect to which Seller is liable without obtaining Seller's prior written consent, which consent shall not be unreasonably withheld or delayed.

(iii) In the event of any conflict or overlap between the provisions of this Section 6.2(g) (Tax Contests), Section 7.5 and Section 7.4, the provisions of this Section 6.2(g) shall control.

(h) Seller and Purchaser agree to furnish or cause to be furnished to each other, upon reasonable request, as promptly as practicable, such information and assistance relating to the Included Businesses, the Purchased Assets and Assumed Liabilities (including reasonable access to books and records) as is reasonably necessary for the preparation and filing of all Tax Returns, the making of any election relating to Taxes, the preparation for any audit by any Taxing Authority, and the prosecution or defense of any action, suit or proceeding, claim, arbitration, litigation or investigation relating to any Tax. Any reasonable expenses incurred in furnishing such information or assistance shall be borne by the Party requesting it.

6.3 <u>Further Assurances</u>. From time to time following the Closing, Seller and Purchaser shall, and shall cause their respective Affiliates to, execute, acknowledge and deliver all such further conveyances, notices, assumptions and releases and such other instruments, and shall take such further actions, as may be commercially reasonably necessary or appropriate to evidence, effectuate and record the sale, transfer, assignment, conveyance and delivery of, and assure fully to, Purchaser and its Designees, successors or assigns, all of the properties, rights, titles, interests, estates, remedies, powers and privileges intended to be sold, transferred, assigned, conveyed and delivered to Purchaser under this Agreement and the Transaction Documents, and to otherwise make effective the transactions contemplated hereby and thereby.

6.4 <u>Business Records</u>. From and after the Closing Date, to the extent reasonably requested by either Party hereto, each Party hereto shall, and shall cause their respective Affiliates to, cooperate with and make available to the other Party, during normal business hours, all Business Records, information and employees (without substantial disruption of employment), as well as, to the extent permitted by the applicable Laws and regulations, access to, and the cooperation of, the auditors of such Party, retained and remaining in existence after the Closing which are necessary or useful in connection with any Tax inquiry, audit, investigation or dispute, the preparation of Tax Returns, the making of any election relating to Taxes, any litigation or investigation or any other matter requiring any such Business Records, information or employees, or access to such auditors, for any reasonable business purpose. The Party requesting any such Business Records, information or employees, or access to such auditors, shall bear all of the out-of-pocket costs and expenses (including, without limitation, attorneys' fees, but excluding any reimbursement for salaries or benefits) reasonably incurred in connection with providing such Business Records, information or employees, or access to such auditors. Any information obtained pursuant to this Section 6.4 or pursuant to any other Section hereof providing for the sharing of information or the review of any Tax Return or other schedule relating to Taxes shall be subject to the Confidentiality Letter. Each of Purchaser and Seller shall retain all books and records with respect to Taxes pertaining to the Purchased Assets for a period of at least four (4) years following the Closing Date.

6.5 <u>Employment Matters</u>. Concurrent with the Closing, Seller shall terminate each Business Employee of Seller that, with Purchaser's approval, has agreed to accept an offer of employment from Purchaser (each, a "<u>Transferred Employee</u>"), effective immediately after the Closing. Seller and its subsidiaries shall pay to such Business Employees of Seller all compensation, bonus and other amounts due and payable to such Business Employees of Seller in connection with such terminations and in accordance with Seller's and its subsidiaries' regular employment policies and practices. For the avoidance of doubt, Purchaser shall have no obligation to hire or retain any Business Employee of Seller, and nothing in this Agreement confers upon any Business Employee of Seller any rights or remedies of any nature or kind whatsoever under or by reason of this Section 6.5.

# 6.6 <u>Miscellaneous Transition Matters</u>.

(a) <u>Collection of Accounts Receivable and Deposits</u>. Subject to Section 6.6(b), Seller agrees that it shall forward promptly to Purchaser any monies, checks or instruments received by Seller or any subsidiary of Seller after the Closing Date with respect to the accounts receivable purchased by Purchaser from Seller or any subsidiary of Seller pursuant to this Agreement. Seller shall provide to Purchaser such reasonable assistance as Purchaser may request with respect to the collection of any such accounts receivable, provided Purchaser pays the reasonable out-of-pocket expenses of Seller and its officers, directors and employees incurred in providing such assistance.

(b) <u>Assumed Contracts In Progress</u>. Notwithstanding Section 6.6(a) above, the parties agree that: (i) Seller shall be entitled to receive, and, if applicable, upon receipt Purchaser shall promptly forward to Seller, all amounts received by Purchaser under the Assigned Contracts listed on Schedule 6.6(b)(i); and (ii) Purchaser shall be entitled to receive, and, if applicable, upon receipt Seller shall promptly forward to Purchaser, all amounts received by Seller under the Assigned Contracts listed on Schedule 6.6(b)(ii).

(c) <u>Use of Compact Platform</u>. One of the three Compact platforms included in the Purchased Assets will be made temporarily available to Seller, solely for Seller's development and demonstration of Seller's NLD technology, on the following terms: Purchaser shall keep a Compact system on its premises, either crated or uncrated in a warehouse or clean room space for a period not to exceed 9 months following the Closing. If still crated, and upon advance request by Seller of not less than 30 days, Purchaser shall uncrate and place in its available clean room space for a period not to exceed 90 days, and on terms and conditions to be agreed between Seller and Purchaser (it being understood that Seller shall pay Purchaser the facilities and administrative cost associated with such display), attach the NLD Chamber as directed by Seller and allow Seller and its representatives reasonable access to the Compact system during normal business hours, solely for the purpose of demonstrating the NLD technology to interested third-parties. If, at the expiration of the 9-month period, Seller has not requested Purchaser to uncrate the Compact System or has not scheduled the showing of the Compact system in the Purchaser's available clean room space, Purchaser shall uncrate the Compact system, dismantle, remove, crate and ship any NLD parts to the Seller (all at Seller's expense).

(d) <u>Name Change of Tegal France</u>. Purchaser shall promptly, but not less than 90 days, effect the change the name of Tegal France to a name that is not associated with Seller or Tegal France and carry out the relevant legal formalities in respect thereto with the French commercial registry.

(e) <u>Appointment of a new director of Tegal France</u>. Purchaser shall promptly after Closing appoint a new director (*Président*) of Tegal France and carry out the relevant legal formalities in respect thereto with the French commercial registry.

(f) <u>Audit Assistance</u>. From and after the Closing, to the extent reasonably requested by Purchaser and at Purchaser's sole cost and expense, in connection with (i) any audit of the financial statements relating to the Included Businesses; (ii) any separate presentation to be prepared by Purchaser of the financial statements relating to the Included Businesses, or (iii) any presentation to be prepared by Purchaser of the *pro forma* effects of Purchaser's acquisition of the Included Businesses, in each case, Seller shall, and shall use commercially reasonable efforts to cause Seller's auditors to, (x) reasonably cooperate in the preparation of such financial statements or pro forma presentation, and (y) provide any records or other information requested by Purchaser in connection therewith.

6.7 <u>Non-Competition</u>. As a material inducement to Purchaser's willingness to enter into and perform this Agreement and to purchase the Purchased Assets and for the consideration to be paid or provided to Seller in connection with such purchase, Seller agrees that it and its subsidiaries will not Compete at any time for three years after the Closing. For purposes of this Section 6.7, "<u>Compete</u>" means directly or indirectly, for its own benefit or as agent for another, participate in the ownership of (except as the passive holder of less than five percent (5%) of the outstanding shares of any class of a corporation whose stock is listed on any national or regional securities exchange), or management or control of, any present or future business enterprise that competes with the Included Businesses, or carry on activities competing with the Included Businesses, as the same was conducted by Seller and its subsidiaries as of the Closing Date.

6.8 <u>Non-Solicitation</u>. From the Closing Date until the third anniversary thereof, Seller shall not solicit or seek to hire away from Purchaser any Transferred Employee or induce any Transferred Employees to terminate his or her employment with Purchaser without Purchaser's consent.

6.9 <u>Alcatel Consent</u>. After the Closing Date, Seller shall use commercially reasonable efforts to assist Purchaser (at Purchaser's expense) in obtaining the consent of Alcatel Lucent, a company incorporated under the laws of France as a *société anonyme*, to the transfer and assignment of the Mark License Contract, dated as of September 2, 2008, by and between Alcatel Lucent and Seller , from Seller to Purchaser.

6.10 <u>Other Consents</u>. After the Closing Date, Seller shall use commercially reasonable efforts to obtain the consent from the counterparties to the Contracts listed on Schedule 6.10.

#### ARTICLE VII

#### INDEMNITY

7.1 <u>Survival</u>. The representations and warranties of the Parties in this Agreement shall survive the Closing Date and be enforceable as set forth in this ARTICLE VII.

7.2 <u>Indemnification by Seller</u>. Subject to Section 7.4 below, Seller shall indemnify, save and hold harmless Purchaser, its Affiliates and their respective officers, directors, agents and representatives (the "<u>Purchaser Indemnified Parties</u>") from and against any and all costs, losses, Taxes, Liabilities, obligations, damages, lawsuits, judgments, settlements, awards, deficiencies, claims, demands, expenses (including reasonable costs of investigation and reasonable attorneys' fees and expenses), interest, fines, penalties, costs of mitigation, and other losses actually paid to third parties (herein, "<u>Damages</u>"), incurred in connection with, arising out of, resulting from or incident to:

(a) any misrepresentation or breach of warranty of Seller contained in this Agreement (other than pursuant to Section 4.2 or Section 4.3(a) of this Agreement) or any certificate delivered pursuant hereto;

(b) any misrepresentation or breach of warranty of Seller contained in Section 4.2 or Section 4.3(a) of this Agreement;

(c) any breach of any covenant or agreement made or to be performed by Seller pursuant to this Agreement; and

(d) any Excluded Liability, other than Excluded Liabilities that are reimbursed or otherwise paid by Seller pursuant to Section 6.2.

7.3 <u>Indemnification by Purchaser</u>. Subject to Section 7.4 below, Purchaser shall indemnify, save and hold harmless Seller, its Affiliates and their respective officers, directors, agents and representatives (the "<u>Seller Indemnified Parties</u>") from and against any and all Damages, incurred in connection with, arising out of, resulting from or incident to:

(a) any misrepresentation or breach of warranty of Purchaser contained in this Agreement (other than pursuant to Section 5.2 of this Agreement) or any certificate delivered pursuant hereto;

(b) any misrepresentation or breach of warranty of Purchaser contained in Section 5.2 of this Agreement;

(c) any breach of any covenant or agreement made or to be performed by Purchaser pursuant to this Agreement;

(d) any Assumed Liability; and

(e) Taxes allocated to Purchaser pursuant to Section 6.2 other than such Taxes that are reimbursed or otherwise paid by Purchaser pursuant to Section 6.2.

## 7.4 <u>Limitations on Indemnification</u>.

(a) Subject to Section 7.4(d), an Indemnifying Party shall not have any liability under Section 7.2(a) or 7.3(a) unless both (i) notice of a claim, specifying the misrepresentation or breach in reasonable detail, is delivered on or before 5:00 p.m., Pacific Time, on the first anniversary of the Closing Date and (ii) the aggregate amount of Damages incurred or to be incurred by the Purchaser Indemnified Parties or Seller Indemnified Parties, as the case may be, pursuant to Section 7.2 or 7.3, as applicable, exceeds \$50,000 (the "Basket") and, in such event, the Indemnifying Party shall be required to pay the entire amount of all such Damages.

(b) Subject to Section 7.4(d), neither Seller nor Purchaser shall be required to indemnify any Person (i) under Section 7.2(a), in the case of Purchaser, or Section 7.3(a), in the case of Seller, for an aggregate amount of Damages exceeding 200,000 (the "Lower Cap") or (ii) under Sections 7.2(a), 7.2(b), 7.2(c), collectively, in the case of Purchaser, or Sections 7.3(a), 7.3(b) or 7.3(c), collectively, in the case of Seller, for an aggregate amount of Damages exceeding the Purchase Price (the "Purchase Price Cap"). For the avoidance of doubt, none of the Damages asserted by Purchaser pursuant to Section 7.2(b), 7.2(c) or 7.2(d), or by Seller pursuant to Section 7.3(b), 7.3(c), 7.3(c) or 7.3(e), regardless of whether asserted pursuant to more than one subsection of Section 7.2 or Section 7.3 (as applicable), shall be included in the calculation of Damages pursuant to Section 7.2(d), or by Seller pursuant to Section 7.3(e), regardless of whether asserted pursuant to Section 7.2(d), or by Seller pursuant to Section 7.3(d) or 7.3(e), regardless of whether asserted pursuant to Section 7.2(d), or by Seller pursuant to Section 7.3(d) or 7.3(e), regardless of whether asserted pursuant to Section 7.2(d), or by Seller pursuant to Section 7.3(d) or 7.3(e), regardless of whether asserted pursuant to Section 7.2(d), or by Seller pursuant to Section 7.3(d) or 7.3(e), regardless of whether asserted pursuant to Section 7.2 or Section 7.3 (as applicable), shall be included in the calculation of Damages pursuant to Section 7.2 or Section 7.3 (as applicable), shall be included in the calculation of Damages pursuant to Section 7.2 or Section 7.3 (as applicable), shall be included in the calculation of Damages pursuant to Section 7.2 or Section 7.3 (as applicable), shall be included in the calculation of Damages pursuant to clause (ii) above for purposes of determining whether the Purchase Price Cap has been exceeded.

(c) Notwithstanding anything in this Agreement to the contrary, for the purpose of this ARTICLE VII, the representations and warranties in this Agreement that are qualified by materiality or Material Adverse Effect or any similar qualification or standard shall be deemed to be made with such qualification or standard for purposes of determining whether a breach of any such representation or warranty has occurred but shall be disregarded for purposes of determining the amount of any related indemnifiable Damages.

- (d) Notwithstanding anything in this Section 7.4, none of the limitations set forth in this Section 7.4 shall apply to:
  - (i) any claim for indemnification based on Sections 7.2(d), 7.3(d) or 7.3(e); or
  - (ii) any claim based upon fraud.

For the avoidance of doubt, claims described in this Section 7.4(d) can be asserted until the expiration of all applicable statutes of limitations, and are not subject to the Basket, the Lower Cap or the Purchase Price Cap.

(e) The amount of Damages recoverable pursuant to this ARTICLE VII shall be net of any Tax benefit realized by the Indemnified Party as a result of the payment or incurrence of such Damages insurance or other recoveries actually received by the Indemnified Party in connection with the facts giving rise to the right of indemnification (but without reduction for any "deductible," "self insurance retention" or any similar risk retention expense in respect of applicable insurance policies or other expenses incurred in connection with such recovery). Nothing herein shall be construed as obligating any Party to maintain any insurance.

7.5 Third-Party Claims. If any lawsuit or enforcement action is filed by a non-Party against any Party entitled to the benefit of indemnity hereunder (the "Indemnified Party"), written notice thereof shall be given to the Party(ies) from whom indemnification will be sought (the "Indemnifying Party") promptly (and in any event within fifteen (15) calendar days after the service of the citation or summons). The failure of any Indemnified Party to give timely notice hereunder shall not affect rights to indemnification hereunder unless and to the extent such failure is prejudicial to the rights or obligations of the Indemnifying Party. After such notice, the Indemnifying Party shall be entitled, if it so elects at its own cost, risk and expense, (i) to take control of the defense and investigation of such lawsuit or action, (ii) to employ and engage attorneys of its own choice to handle and defend the same unless the named parties to such action or proceeding include both the Indemnifying Party and the Indemnified Party and the Indemnified Party has been advised in writing by counsel that there may be one or more legal defenses available to such Indemnified Party that are different from or additional to those available to the Indemnifying Party, in which event the Indemnified Party shall be entitled, at the Indemnifying Party's cost, risk and expense, to separate counsel of its own choosing, and (iii) to compromise or settle such claim, which compromise or settlement shall be made only with the written consent of the Indemnified Party, such consent not to be unreasonably withheld or delayed;. In the event the Indemnifying Party assumes the defense of the claim, the Indemnifying Party will keep the Indemnified Party reasonably informed of the progress of any such defense, compromise or settlement. If the Indemnifying Party fails to assume the defense of such claim within thirty (30) calendar days after receipt of notice of the claim, the Indemnified Party against which such claim has been asserted will (upon delivering notice to such effect to the Indemnifying Party) have the right to undertake, at the Indemnifying Party's cost and expense, the defense, compromise or settlement of such claim on behalf of and for the account and risk of the Indemnifying Party; provided, however, that such claim shall not be compromised or settled without the written consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed. In the event the Indemnified Party assumes the defense of the claim, the Indemnified Party will keep the Indemnifying Party reasonably informed of the progress of any such defense, compromise or settlement.

# 7.6 <u>Escrow</u>.

(a) Any amount that Seller is obligated to pay to a Purchaser Indemnified Party pursuant to this ARTICLE VII shall, to the extent there are sufficient Escrow Funds available at the time such payment is due, be paid first by the disbursement of Escrow Funds to such Purchaser Indemnified Party in accordance with the terms of Escrow Agreement.

(b) On the date that is the first Business Day following the first anniversary of the Closing Date, the Escrow Agent shall release to Seller the balance of Escrow Funds available at such time, *provided* that the Escrow Agent shall retain an amount (up to the total amount of Escrow Funds then available) equal to the aggregate amount of Damages asserted in good faith by Purchaser Indemnified Parties in claims for indemnification under this ARTICLE VII that were asserted prior to, and remain unresolved as of, such date (the "<u>Unresolved Claims</u>"). The portion of the Escrow Funds so retained for each Unresolved Claim shall be released by the Escrow Agent (to Seller, the Purchaser Indemnified Party asserting the claim or both, as applicable) only upon resolution of such Unresolved Claim and in accordance with the terms of the Escrow Agreement.

7.7 <u>Tax Effect on Indemnification Payments</u>. All indemnity payments made by Seller to Purchaser Indemnified Parties, or by Purchaser to Seller Indemnified Parties, pursuant to this Agreement shall be treated for all Tax purposes as adjustments to the Purchase Price.

7.8 Exclusive Remedy. Following the Closing, this ARTICLE VII shall provide the sole and exclusive remedy of an Indemnified Party for any Damages arising under Sections 7.2 or 7.3 as applicable (and as so limited herein); provided, however, that this Section 7.8 shall not apply, and an Indemnified Party shall be entitled to pursue any remedy available at Law or in equity, with respect to any Damages incurred or suffered by an Indemnified Party arising from or related to fraud; provided, further, that nothing in this Section 7.8 or elsewhere in this Agreement shall affect the parties' rights to specific performance or other equitable remedies that do not involve the payment or other monetary compensation from the Indemnifying Party to the Indemnified Party with respect to the covenants and agreements in this Agreement that are to performed at or after the Closing. In the case of fraud, all rights and remedies existing under this Agreement will be cumulative to, and not exclusive of, any rights or remedies otherwise available to the injured party, and the rights and remedies set forth in this ARTICLE VII will not be deemed to preclude or otherwise limit in any way the exercise of any other rights or pursuit of any other remedies.

## ARTICLE VIII

# MISCELLANEOUS

8.1 <u>Expenses</u>. Each party hereto shall pay all of its own fees, costs and expenses (including, without limitation, those of advisors, financial advisors, lawyers or accountants) incurred by it in connection with the negotiation, preparation, execution, delivery and performance of this Agreement and the transactions contemplated hereby.

8.2 <u>Notices</u>. All notices and other communications provided for herein shall be in writing and shall be deemed to have been duly given when delivered personally or when sent by telecopy or other electronic or digital transmission method (including, but not limited to, in portable document format by electronic mail) or three (3) business days after being mailed by registered or certified mail, return receipt requested, postage prepaid, to the party to whom it is directed or one (1) Business Day after being sent via an internationally recognized courier service for next business day delivery, to the party to whom it is directed:

If to Purchaser, to:

SPP Process Technology Systems, Ltd. Ringland Way Newport NP18 2TA United Kingdom Attention: Richard Rees, VP & Chief Finance Officer Facsimile: +44.1633.414141 With a copy to:

Bingham McCutchen LLP 1900 University Avenue East Palo Alto, California 94303 Attention: Bart Deamer Facsimile: +1.650.849.4800

If to Seller, to:

Tegal Corporation 2201 S. McDowell Blvd. Petaluma, CA 94954 Attention: Thomas R. Mika Facsimile: +1.707.763.0415

With a copy to:

Latham & Watkins LLP 140 Scott Drive Menlo Park, California 94025 Attention: Robert Phillips Facsimile: +1.650.463.2600

or for any party, at such other address as such party shall have specified in writing to each of the others in accordance with this Section 8.2.

8.3 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but such counterparts together shall constitute one and the same instrument.

8.4 <u>Section Headings</u>. The section headings of this Agreement are for convenience of reference only and shall not be deemed to limit or affect any of the provisions hereof.

## 8.5 <u>Amendments; No Waivers.</u>

(a) Any provision of this Agreement may be waived or amended if, and only if, such amendment or waiver is in writing and signed by each of the parties hereto.

(b) No failure by either Party hereto to insist upon the strict performance of any covenant, duty, agreement or condition of this Agreement, or to exercise any right or remedy consequent upon a breach hereof, shall constitute a waiver of any such breach or any other covenant, duty, agreement or condition hereof.



8.6 <u>Entire Agreement; No Assignment</u>. This Agreement (including the Exhibits hereto, Disclosure Letter and any amendments hereto), the other Transaction Documents and the Confidentiality Agreement constitute the entire agreement and understandings of the parties hereto and supersedes all prior agreements and understandings, both written and oral, among the parties hereto with respect to the subject matter. This Agreement is not intended to confer upon any other Person any rights or remedies hereunder and shall not be assigned, by operation of Law or otherwise, prior to the Closing, except that Purchaser may assign its rights and delegate its obligations hereunder to one or more Designees, to the extent such rights and obligations pertain to the Purchased Assets and Assumed Liabilities that such Designee has been designated to receive and assume.

8.7 <u>No Third Party Beneficiaries</u>. Neither any Seller nor Purchaser intends for this Agreement or the other Transaction Documents to create any rights or obligations except as between Seller and Purchaser, and no other Person shall be treated as third-party beneficiaries hereunder or thereunder, except that each Designee designated by Purchaser to acquire any Purchased Assets pursuant to Section 2.1 shall be deemed to be a beneficiary of Seller's covenants hereunder with respect to such Purchased Assets.

8.8 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the Laws of the State of California (without regard to the choice of Law provisions thereof).

8.9 <u>Severability</u>. If it is determined by a court of competent jurisdiction that any provision of this Agreement is invalid under applicable Law, such provision shall be ineffective only to the extent of such invalidity, without invalidating the remainder of this Agreement.

8.10 <u>Specific Performance; Preservation of Remedies</u>. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with its specific terms or were otherwise breached. It is agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of covenants in this Agreement and to enforce specifically the terms and provisions hereof, this being in addition to any other remedy to which they are entitled at Law or in equity. Except as otherwise provided herein, all rights and remedies of either party hereto are cumulative of each other and of every other right or remedy such party may otherwise have at Law or in equity, and the exercise of one or more rights or remedies shall not prejudice or impair the concurrent or subsequent exercise of other rights or remedies.

[signature page follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed as of the date hereof.

SPP PROCESS TECHNOLOGY SYSTEMS LTD.

By: /s/ William Johnson Name: William Johnson Title: President & CEO

TEGAL CORPORATION

By: /s/ Thomas R. Mika Name: Thomas R. Mika Title: President & CEO

# SIGNATURE PAGE TO ASSET PURCHASE AGREEMENT

#### TRADEMARK LICENSE AGREEMENT

This **Trademark License Agreement** ("*Agreement*") is effective as of the Closing Date by and between **Tegal Corporation**, a Delaware corporation ("*Seller*") and **SPP Process Technology Systems Ltd.**, a company incorporated and registered in England and Wales ("*Purchaser*").

WHEREAS, Seller and Purchaser have entered into an Asset Purchase Agreement, dated as of February 9, 2011 (the "Purchase Agreement"); and

**WHEREAS**, in connection with the foregoing, Seller desires to grant to Purchaser and its Designees a license to use certain Licensed Marks (as defined below);

**NOW, THEREFORE**, in consideration of the mutual promises of the parties hereto, and of good and valuable consideration, it is agreed by and between the parties as follows:

1. **DEFINITIONS**. For the purposes of this Agreement the following capitalized terms are defined in this Section 1 and shall have the meaning specified herein. Other terms that are capitalized but not specifically defined in this Section 1 or in the body of the Agreement shall have the meanings set forth in the Purchase Agreement.

1.1. "Designee" means the Affiliates of Purchaser designated pursuant to the Designation Notice ...

1.2. *"Licensed Marks*" means the trademarks of Seller set forth in Exhibit A attached hereto.

1.3. "*Purchaser Products*" means any products designed, developed, distributed, marketed, manufactured, tested, assembled, used, imported, licensed, offered for sale, and/or sold in connection with the Included Business.

1.4. *"Trademark Usage Guidelines"* means the written guidelines for proper usage of the Licensed Marks, as in use by Seller immediately prior to the Closing Date and as revised and updated from time to time by Seller with notice to Purchaser. All such guidelines may be revised and updated by Seller from time to time during the term of the license in its reasonable discretion.

1.5. *"Transferred Trademarks"* means the trademarks transferred to Purchaser and its Designees as part of the Purchased Intellectual Property pursuant to the Purchase Agreement.

# 2. LICENSES.

2.1. **License Grant**. Subject to the terms and conditions of this Agreement, Seller hereby grants to Purchaser and each of its Designees an exclusive, non-transferable (except as permitted pursuant to Section 7.5), worldwide, royalty-free license, without right of sublicense, during the term of this Agreement to use, copy, distribute and reproduce the Licensed Marks solely in connection with the manufacture, assembly, use, import, export, design, development, distribution, marketing, selling and support of the Purchaser Products and solely in combination with the Transferred Trademarks (for example, "Tegal  $4200SE^{TM*}$ ). Purchaser guarantees the compliance of each of its Designees with the terms and conditions of this Agreement.

2.2. License Restrictions. Purchaser agrees to, and to cause its Designees to, use the Licensed Marks during the term of the license only in a manner that is consistent with the Trademark Usage Guidelines that Seller may furnish to Purchaser from time to time, including all required trademark notices. Purchaser agrees not to, and will cause its Designees not to, take any action inconsistent with such ownership and further agrees to take, and to cause its Designees to take, at Seller's reasonable expense, any action, including the conduct of legal proceedings, which Seller deems necessary to establish and preserve Seller's rights in and to the Licensed Marks. Purchaser shall not, and shall cause its Designees not to, adopt, use or attempt to register any trademarks or trade names that are confusingly similar to the Licensed Marks or in such a way as to create combination marks with the Licensed Marks. Upon Seller's request, Purchaser shall provide, or shall cause its Designees to provide, Seller with samples of all materials that use the Licensed Marks prior to their public use or display for Seller's quality control purposes.

3. OWNERSHIP OF LICENSED MARKS. Purchaser acknowledges Seller's exclusive ownership of the Licensed Marks and the renown of the Licensed Marks worldwide. As between the parties, Seller shall own all right, title and interest in the Licensed Marks by Purchaser and its Designees shall inure exclusively to the benefit of Seller, and neither Purchaser nor any of its Designees shall acquire or assert any rights therein. Purchaser recognizes the value of the goodwill associated with the Licensed Marks, and that the Licensed Marks may have acquired secondary meaning in the minds of the public.

# 4. [RESERVED].

## 5. TERM; TERMINATION.

5.1. **Term**. Unless otherwise agreed by the parties or as set forth in Section 5.2 below, this Agreement will be effective on a perpetual basis.

5.2. **Termination**. Seller may terminate this Agreement if Purchaser materially breaches a provision of this Agreement and fails to cure such breach within thirty (30) days after Seller provides written notice of such breach to Purchaser.

5.3. **Survival.** Sections 3, 5.3, 6 and 7 shall survive any termination or expiration of this Agreement or the licenses hereunder.

# 6. LIMITATION OF LIABILITY.

TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, IN NO EVENT SHALL EITHER PARTY OR ITS AFFILIATES BE LIABLE TO THE OTHER PARTY, ITS AFFILIATES OR TO ANY THIRD PARTY CLAIMING THROUGH OR UNDER SUCH OTHER PARTY, FOR ANY LOST PROFITS OR FOR ANY SPECIAL, CONSEQUENTIAL, INDIRECT OR INCIDENTAL DAMAGES, HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY (INCLUDING NEGLIGENCE) ARISING IN ANY WAY OUT OF THIS AGREEMENT, WHETHER OR NOT SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

# 7. MISCELLANEOUS.

7.1. **No Obligation to Obtain or Maintain Marks**. Neither party shall be obligated to: (a) file any application for registration of any trademark, or to secure any rights in any trademarks, (b) maintain any trademark registration, or (c) provide any assistance, except for the obligations expressly assumed in this Agreement.

7.2. **Amendment and Modification**. This Agreement may be amended, modified or supplemented, only by a written agreement signed by the parties hereto.

7.3. **Waiver of Compliance; Consents**. Any failure of Purchaser, on the one hand, or Seller, on the other hand, to comply with any obligation, covenant, agreement or condition herein may be waived by Seller or Purchaser, respectively, only by a written instrument signed by the party granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure. Whenever this Agreement requires or permits consent by or on behalf of a party, such consent shall be given in writing in a manner consistent with the requirements for a waiver of compliance as set forth in this Section 7.3.

7.4. **Notices.** All notices and other communications provided for herein shall be in writing and shall be deemed to have been duly given when delivered personally or when sent by telecopy or other electronic or digital transmission method (including, but not limited to, in portable document format by electronic mail) or three (3) business days after being mailed by registered or certified mail, return receipt requested, postage prepaid, to the party to whom it is directed or one (1) business day after being sent via an internationally recognized courier service for next business day delivery, to the party to whom it is directed:

If to Purchaser, to: SPP Process Technology Systems, Ltd. Ringland Way Newport NP18 2TA United Kingdom Attention: Richard Rees, VP & Chief Finance Officer Facsimile: +44.1633.414141 With a copy to:

Bingham McCutchen LLP 1900 University Avenue East Palo Alto, California 94303 Attention: Bart Deamer Facsimile: +1.650.849.4800

If to Seller, to:

Tegal Corporation 2201 S. McDowell Blvd. Petaluma, CA 94954 Attention: Thomas R. Mika, President and Chief Executive Officer Facsimile: +1.707.763.0415

With a copy to:

Latham & Watkins LLP 140 Scott Drive Menlo Park, California 94025 Attention: Robert Phillips Facsimile: +1.650.463.2600

or for any party, at such other address as such party shall have specified in writing to each of the others in accordance with this Section 7.4.

7.5. **Assignment**. Seller may assign this Agreement without Purchaser's consent upon any dissolution, liquidation, reorganization, change of control, merger, acquisition, or sale of all or substantially all of the assets of Seller. Except as expressly set forth herein, Purchaser may not assign or transfer this Agreement, directly or indirectly, in whole or in part, whether voluntarily or involuntarily or by operation of law or otherwise, without Seller's prior written consent. Any assignment in violation of this Section 7.5 shall be null and void from the beginning; provided, however, that Purchaser may assign this Agreement or any or all of its rights and interests hereunder to one or more of its affiliates or upon any change of control, merger, acquisition or sale of substantially all of the assets of Purchaser.

7.6. **Governing Law**. This Agreement shall be governed by and construed in accordance with the domestic laws of the State of California, without giving effect to any choice of law or conflict of law provision or rule that would cause the application of the laws of any jurisdiction other than the State of California.

7.7. **Interpretation**. The section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the parties and shall not in any way affect the meaning or interpretation of this Agreement. The parties hereto are sophisticated, represented by counsel and jointly have participated in the negotiation and drafting of this Agreement and there shall be no presumption or burden of proof favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

7.8. **No Third Party Beneficiaries**. This Agreement is not intended to, and does not, create any rights or benefits of any party other than the parties hereto.

7.9. **Severability**. Wherever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but in case any one or more of the provisions contained herein shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provision of this Agreement, and this Agreement shall be construed as if such invalid, illegal or unenforceable provision or provisions had never been contained herein unless the deletion of such provision or provisions would result in such a material change as to cause completion of the transactions contemplated hereby to be unreasonable.

7.10. **Construction**. Unless the context of this Agreement otherwise requires: (i) words of any gender include each other gender; (ii) words using the singular or plural number also include the plural or singular number, respectively; (iii) the terms "hereof," "herein," "hereby" and derivative or similar words refer to this entire Agreement; (iv) the term "Section" refers to the specified Section of this Agreement; (v) the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or"; and (vi) the term "including" means "including without limitation". Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified.

7.11. **Counterparts**. This Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original and all of which together shall be deemed to be one and the same instrument. Copies of executed counterparts transmitted by telecopy, telefax or other electronic transmission service shall be considered original executed counterparts for purposes of this Section 7.11, provided that receipt of copies of such counterparts is confirmed.

7.12. **Entire Agreement**. This Agreement (including Exhibit A hereto) and the Purchase Agreement constitute the entire understanding among the parties with respect to the subject matter hereof and shall supersede all prior written and oral and all contemporaneous oral agreements and understandings with respect to the subject matter hereof.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties have signed this Trademark License Agreement effective as of the Closing Date first set forth above.

# SPP PROCESS TECHNOLOGY SYSTEMS LTD.

By:	William Johnson
Name:	William Johnson
Title:	President & CEO
TEGAL	CORPORATION
By:	Thomas R. Mika
Name:	Thomas R. Mika
Title:	President & CEO

# EXHIBIT A

# LICENSED MARKS

(see attached)

# Exhibit A

# **TEGAL CORPORATION**

Portfolio #	Trademar	k Country	App Date	App #	Reg Date	Reg #	Renewal	Status
1067CN0	TEGAL (d.	7) China	5/28/2008	PENDING 6750532				Publ 1-16-10
1067CN1	Tegal (cl.9		11/12/2008	7053079				Pending
1067MY0	TEGAL	Malaysia	5/23/2008	08010093				Pending
				REGISTERED				
1067CT0	TEGAL	CTM (EU)	6/23/1998	858043	4/17/2000	858043	6/23/2018	Registered
1067JP0	TEGAL	Japan	5/22/2008	2008-039193	6/5/2009	5236399	6/5/2019	Registered
1067KR0	TEGAL	Korea	5/21/2008	40-2008-0024581	5/20/2009	40-0789505	5/20/2019	Registered
1067TW0 1067US0	TEGAL TEGAL	Taiwan US	5/23/2008 6/26/1987	097024586 73/668,774	5/1/2009 5/1/2009	1361449 1,494,896	5/1/2019 7/5/2018	Registered Registered
1105CT0	Tegal w/arro		6/23/1998	858993	7/5/1988	858993	6/23/2018	Registered
<u>Non-register</u> ProNova2 ProNova IP1	red Trademarks Source Source Source							
IP1.5	Source							
IP2	Source							
IP3	Source							
IP4	Source							
	110 Platform							
		Platform						
	4200	Platform						
	3200	Platform						
SHARP		Process						



Tegal Corporation 2201 S. McDowell Blvd., Petaluma, CA 94954 [T] (707) 763-5600 [F] (707) 765-9311 www.tegal.com

# TEGAL SELLS DRIE ASSETS TO SPTS

#### Tegal's focus now exclusively on Green Energy

**PETALUMA, Calif., February 9, 2011**—<u>Tegal Corporation</u>, (Nasdaq: TGAL) today announced the sale of its Deep Reactive Ion Etch (DRIE) and certain related assets to SPP Process Technology Systems Ltd (SPTS). Tegal will receive aggregate consideration of approximately \$2.1 million, including \$1.6 million in cash and the assumption by SPTS of approximately \$0.5 million of short-term liabilities. The deal includes the transfer to SPTS of the capital stock and operations of Tegal France SAS, a wholly-owned Tegal subsidiary formed following the acquisition in September 2008 by Tegal of the DRIE assets of Alcatel Micro Machining Systems.

Tegal sold to SPTS the Tegal 200<sup>TM</sup>, 110<sup>TM</sup>, 3200<sup>TM</sup> and 4200<sup>TM</sup> series DRIE products, along with its Compact<sup>TM</sup> and Pluto development assets, intellectual property and process know-how. SPTS will provide continued global support to existing Tegal DRIE customers, and will integrate these technologies into its broad range of product offerings in etch, deposition and thermal technologies which it offers to its customers worldwide.

"The sale of the DRIE product lines to SPTS represents another major step in Tegal's transformation. We recently announced our participation in the formation of sequel Power, a company dedicated to the development and operation of large scale photovoltaic (PV)-based solar utilities in the United States, Latin America, the Middle East and Africa. From this platform, we intend to build additional operations and make further investments as we transition from being a supplier of semiconductor capital equipment to pursuing a leading role in green energy," said Thomas Mika, President and CEO, Tegal Corporation. "The sale of the DRIE assets to SPTS ensures a smooth transition of our technology, team members, active joint development projects in France, and our commitments to customers globally."

"This acquisition enhances the DRIE product portfolio SPTS offers to our served markets, broadens our customer base, and expands our ability to deliver world class DRIE solutions to our global customers," said William Johnson, President and Chief Executive Officer of SPTS. "We expect to enhance support to Tegal customers through our worldwide support network, and intend to continue the leading-edge development efforts of the former Tegal team in Annecy, France.

#### Safe Harbor Statement

Except for historical information, matters discussed in this news release contain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Forward-looking statements, which are based on assumptions and describe our future plans, strategies and expectations, are generally identifiable in some instances by the use of the words "anticipate," "believe," "estimate," "expect," "intend," "project" or similar expressions. These forward-looking statements are subject to risks, uncertainties and assumptions, including, but not limited to industry conditions, economic conditions, acceptance of new technologies and market acceptance of products and services. All forward-looking statements attributable to Tegal Corporation or persons acting on its behalf are expressly qualified in their entirety by the cautionary statements in this paragraph. For a further discussion of these risks and uncertainties, please refer to Tegal's periodic filings with the Securities and Exchange Commission.

### **About Tegal Corporation**

Tegal is dedicated to the development and application of both proven and emerging technologies in the field of green energy. Through sequel Power, Tegal is engaged in the promotion of solar power plant development projects worldwide, the development of self-sustaining businesses from such projects, including supporting, developing, building and operating solar photovoltaic fabrication facilities and solar farms and other non-PV based renewable energy projects. Headquartered in Petaluma, California, the company has more than 35 years of semiconductor capital equipment expertise and innovation in specialized technologies. Please visit us on the web at <u>www.Tegal.com</u>.

# About SPP Process Technology Systems

SPP Process Technology Systems was established in October 2009 as the vehicle for the merger of Surface Technology Systems and acquired assets of Aviza Technology. The company is a wholly-owned subsidiary of Sumitomo Precision Products Co., Ltd, and designs, manufactures, sells, and supports advanced semiconductor capital equipment and process technologies for the global semiconductor industry and related markets. These products are used in a variety of market segments, including R&D, data storage, MEMS and nanotechnology, advanced 3-D packaging, LEDs, and power integrated circuits for communications. For more information about SPTS, please visit <u>www.spp-pts.com</u>

-

**Tegal Contact** Thomas Mika

President & CEO Tel: +1 707 765-5630 Email: <u>tmika@tegal.com</u> or The Blueshirt Group Chris Danne Tel: +1 415 217-7722

# **SPTS Contact**

Evelyn Tay, Corporate Marketing SPP Process Technology Systems Ph: (+65) 8383 0393 Email: <u>evelyn.tay@spp-pts.com</u>

##